

IBM Cúram Social Program Management

## Cúram Integrated Case Management Guide

Version 6.0.4

<b>Note</b> Before using this information and the product it supports, read the information in Notices at the back of this guide.
This edition applies to version 6.0.4 of IBM Cúram Social Program Management and all subsequent releases and modifications unless otherwise indicated in new editions.
Licensed Materials - Property of IBM
Copyright IBM Corporation 2012. All rights reserved.
US Government Users Restricted Rights - Use, duplication or disclosure restricted by GSA ADP Schedule Contract with IBM Corp.
© Copyright 2011 Cúram Software Limited

## **Table of Contents**

Chapter 1 Introduction	1
1.1 Purpose	1
1.2 Audience	
1.3 Prerequisites	1
1.4 Chapters in this Guide	
•	
Chapter 2 Needs to Outcomes and Integrated Case Management	3
2.1 Introduction	
2.2 Needs to Delivery - Cúram Integrated Case Management	3
2.2.1 Delivering Benefits through Product Deliveries	4
2.2.2 Meeting Needs through Service Deliveries	5
2.2.3 Meeting Needs through Referrals	6
2.3 Cúram Outcome Management - Assessments to Outcomes	6
2.4 The Cúram Integrated Case - A Holistic View	7
2.5 Helping Case Workers Do Their Jobs - The Case Worker Workspace	7
2.6 Achieving Sustainable Outcomes Through Ongoing Case Management	8
2.7 Supporting Agency Operations Through Configuration	
Chapter 3 Assessing Needs - From Initial Contact to Integrated Case Creation	
3.1 Introduction	
3.2 Supporting the Agency's Triage, Screening, and Intake Processes	
3.3 Triaging Clients for Programs and Services	
3.4 Screening Clients for Programs	
3.5 Program Intake Processes	
3.6 Creating an Integrated Case	
3.6.1 Assigning Initial Case Ownership	
3.6.2 Integrated Case Information	15
	1.7
Chapter 4 Delivering Benefits to Clients through Product Delivery Cases	
4.1 Introduction	
4.2 Creating a Product Delivery Case	
4.3 Capturing Real World Information about a Client	
4.3.1 Capturing Evidence Using the EvidenceFlow View	
4.3.2 Identifying Concerns with Evidence that Affect Eligibility	
4.4 Checking Eligibility for Active and In Edit Evidence	
4.5 Approving Payments and Activating a Case	
4.5.1 Submitting a Case for Approval	20

4.5.2 Approving or Rejecting a Case	21
4.5.3 Activating a Case	
4.6 Applying Rules to Real World Information to Make Decisions	
4.7 Reading Decisions, Eligibility, Entitlement, and Explanation	
4.7.1 Viewing Decisions over the Lifetime of a Case	
4.7.2 Comparing Decisions	
4.8 Scheduling Payments and Bills- the Financial Scheduler	
4.8.1 Case Nominees and Case Components	
4.8.2 Scheduling Payments and Bills	
4.8.3 Payment and Billing Processing	
4.8.4 Issuing Payments and Bills	
4.8.5 Setting up Case Deductions	
4.8.6 Simulating Payments	
4.8.7 Viewing Financial Transactions	
4.9 Monitoring Recent Case Transactions	
4.9.1 Monitoring Recently Assigned, Approved, and Viewed Cases	
4.9.2 Viewing Related Cases	
4.9.3 Using the Case Transaction Log	
4.9.4 Understanding the Case Status History	
1.5.1 Chacistanding the Case Status History	52
Chapter 5 Delivering Services to Clients through Service Delivery and Referrals	34
5.1 Introduction	
5.2 Creating a Service for a Client	34
5.2.1 Specifying an Appropriate Service for a Client	35
5.2.2 Assigning Service Ownership	35
5.2.3 Recommending the Required Number of Units of a Service	35
5.2.4 Authorizing the Provider Rate	36
5.2.5 Recommending the Frequency of the Service	36
5.2.6 Indicating the Client's Required Participation in the Service	36
5.2.7 Specifying a Nominee to Receive Payments in Respect of the Service	36
5.2.8 Specifying a Provider and Provider Type	37
5.3 Referring a Client to a Service Provider	37
5.3.1 Specifying a Referral Service	37
5.3.2 Recording the Referral Date	37
5.3.3 Specifying to Follow Up With a Client	38
5.3.4 Sending Notification Letters to the Service Provider and Client	38
5.4 Checking a Client's Schedule and Availability to Attend a Service	
5.5 Finding the Best Service Provider to Meet a Client's Needs	39
5.5.1 Locating the Nearest Service Provider	40
5.5.2 Validating Provider Credentials	
5.5.3 Estimating the Cost of a Service	41
5.6 Enquiring About a Service and Recording the Provider Response	42
5.7 Approving a Service Delivery	42
5.7.1 Submitting a Service Delivery for Approval	
5.7.2 Approving or Rejecting a Service Delivery	
5.7.3 Viewing Service Approval Requests	44
5.8 Paying Clients and Service Providers	
5.8.1 Processing Payments	
5.8.2 Viewing Financial Transactions in Respect of a Service	45

5.8.3 Determining the Actual Cost of the Service	
5.9 Making Changes to a Service	46
5.10 Completing a Service and Recording an Outcome	47
5.11 Evaluating the Quality of a Service	47
5.12 Viewing the Service Delivery Status History	
Chapter 6 The Case Worker Workspace	
6.1 Introduction	
6.2 Accessing Caseload	
6.3 Viewing Daily Workload Summaries	51
6.4 The Case Calendar	
6.5 Searching and Querying Cases	
6.5.1 Searching for Specific Case Information	53
6.5.2 Performing a Quick Search by Reference Number	54
6.5.3 Monitoring Cases Using Case Queries	54
6.6 Recording Items of Interest	54
Chapter 7 Ongoing Case Management	
7.1 Introduction	
7.2 Manually Changing the Case Owner and Case Supervisor	
7.3 Viewing Contextual Information	
7.3.1 Viewing Contextual Information for Cases	
7.3.2 Viewing Contextual Information for Integrated Cases	58
7.3.3 Viewing Contextual Information for Product Delivery Cases	
7.4 Auditing Cases	
7.5 Using The Case Calendar to Schedule Meetings and Track Case Events	
7.5.1 Scheduling Meetings and Recording Meeting Minutes	59
7.5.2 Scheduling Activities	
7.5.3 Referring Clients for Services	
7.5.4 Setting up Case Reviews	
7.5.5 Appealing Services	
7.5.6 System Generated Events	
7.6 Working with Case Tasks	
7.7 Ongoing Client Management	
7.7.1 Managing Case Members and Participants	
7.7.2 Determining the Need for a Translator	
7.7.3 Maintaining Client Contracts	
7.7.4 Maintaining Communications	
7.7.5 Adding Attachments	
7.8 Recording Notes	65
7.9 Certifying Case Evidence	
7.10 Tracking Issues and Legal Proceedings	
7.10.1 Issues	
7.10.2 Investigations	67
7.10.3 Appeals	
7.10.4 Legal Actions	
7.10.5 Special Cautions	
7.11 Reassessing Case Eligibility and Entitlement	69
7.11.1 Over and Under Payment Processing	
7.12 Suspending a Case	70

7.13 Closing a Case	. 70
7.14 Reactivating a Case	
Charten O Car Carrier Internated Care Management	70
Chapter 8 Configuring Integrated Case Management	. 12
8.1 Introduction	
8.2 Configuring Case Types	
8.3 Configuring Products	
8.3.1 Configuring General Product Information	
8.3.2 Configuring Eligibility Determination Processing	
8.3.3 Configuring Delivery Patterns	
8.3.4 Configuring Product Provisions and Assigning them to Locations	
8.3.5 Configuring Benefit Deduction Information	
8.3.6 Configuring Liability Information	
8.3.7 Configuring Case Reassessment and Adjustments	
8.3.8 Configuring Certification Requirements	
8.3.9 Configuring Bonus Payments	
8.3.10 Configuring Absence Reasons	. 77
8.3.11 Setting Product Time Constraints	
8.3.12 Configuring Payment Corrections	
8.3.13 Configuring Rate Tables	
8.3.14 Applying System Wide Changes to Cases	
8.3.15 Configuring Payment Groups	
8.4 Configuring Case Evidence	
8.4.1 Configuring Evidence Types - The Dynamic Evidence Editor	
8.4.2 Activating Dynamic Evidence	
8.4.3 Associating Evidence Types with Cases	. 81
8.4.4 Setting Evidence Approval Checks	. 81
8.4.5 Configuring the EvidenceFlow View	
8.5 Defining Rules and Assigning Them to Products	. 82
8.5.1 Dynamic Rules Design - The CER Editor	. 82
8.5.2 Linking Rules to Evidence	. 83
8.5.3 Assigning Rules to Products	. 83
8.6 Configuring Services and Referrals	. 84
8.6.1 Configuring Services	
8.6.2 Configuring Referral Services	. 86
8.7 Configuring Case Ownership	. 87
8.8 Configuring Case Milestones	
8.9 Configuring Ongoing Case Management	
	0.0
Chapter 9 Securing Cases	
9.1 Introduction	
9.2 Securing General Case Operations	. 90
9.3 Protecting Fields from General User Access	
9.4 Securing Case and Client Information Based on Location	
9.5 Securing Operations for Specific Programs and Features	
9.6 Securing Access to Sensitive Case Information	. 92
Chapter 10 Conclusion	03
10.1 Summary	
10.1 Summary	. 93

	Cúram	Integrated	Case	Manag	ement	Guide
--	-------	------------	------	-------	-------	-------

## Chapter 1

## Introduction

## 1.1 Purpose

The purpose of this guide is to provide a business overview of Cúram Integrated Case Management and where this fits into the application's 'Needs to Outcomes' approach. Cúram Integrated Case Management works in conjunction with Cúram Outcome Management<sup>TM</sup> by providing a holistic approach to assessing the needs of families and meeting these needs through delivery of programs and services and through outcome management.

Cúram Integrated Case Management focuses on assessing needs and delivering programs and services to meet these needs; Outcome Management assesses needs and designs plans to achieve sustainable outcomes.

Although this guide makes reference to Outcome Management, this guide primarily focuses on the needs to delivery process. The needs to delivery process is made up of two main parts: meeting needs through the delivery of programs in the form of benefits and meeting needs through service deliveries and referrals.

### 1.2 Audience

This guide is intended for any reader interested in understanding the business concepts of Cúram Integrated Case Management. Readers do not require knowledge of the application.

## 1.3 Prerequisites

To better understand the topics covered in this guide, it is suggested that the reader has previously read the Cúram Participant Guide.

## 1.4 Chapters in this Guide

The following list describes the chapters within this guide:

#### **Needs to Outcomes and Integrated Case Management**

This chapter introduces Cúram Integrated Case Management and describes how it supports the 'Needs to Outcomes' approach.

#### **Assessing Needs - From Initial Contact to Integrated Case Creation**

The chapter provides an overview of the process of initial contact between a client and the agency, the assessment of a client's needs using triage, screening and intake, and the creation of an integrated case to manage programs and services that meet the needs of clients.

#### **Delivering Benefits Through Product Delivery Cases**

This chapter provides an overview of the business processes that allow the agency to deliver benefits to clients. The processes include creating a product delivery case, capturing real world information about the client, client eligibility and entitlement determination, and the delivery of the benefit to the client in the form of monetary payments.

#### **Delivering Services Through Service Deliveries and Referrals**

This chapter provides an overview of the business processes that allow the agency to deliver services to clients. These processes support the agency in providing its own services to clients as well as services that are provided by external providers and paid for by the agency. Cúram Integrated Case Management also supports client referrals to services that are not paid for by the agency.

#### The Case Worker Workspace

This chapter provides an overview of the main elements of the case worker workspace. The case worker workspace allows case workers with heavy case loads to manage their daily work effectively. Key elements of the case worker workspace include tools for accessing case load, searching for cases, and the case calendar.

#### **Ongoing Case Management**

This chapter provides information on the tools that support the agency's ongoing case management process. Ongoing case management includes managing case participants, auditing cases, managing and reassessing case eligibility and entitlement, as well as tracking issues and legal proceedings.

#### **Configuring Integrated Case Management**

This chapter provides an overview of the main aspects of Cúram Integrated Case Management that are configured as part of the application administration. This includes configuring case types, the rules and evidence types used to determine client eligibility and entitlement, as well as various aspects of ongoing case management. Services and referral services are also fully configurable.

#### **Securing Cases**

This chapter provides an overview of the options available for securing case and client information.

## Chapter 2

## Needs to Outcomes and Integrated Case Management

#### 2.1 Introduction

Needs to Outcomes offers a two-pronged approach that delivers care and protection to meet the needs of clients as they work towards a desired outcome. Cúram Integrated Case Management supports 'Needs to Outcomes' by enabling agencies to deliver products and services to clients while also helping them to achieve self sufficiency.

Cúram Integrated Case Management works in conjunction with Cúram Outcome Management to provide a cost-effective, end-to-end set of business processes that facilitate agencies in helping clients achieve self sufficiency and reach their socio-economical potential no matter what their situation. This set of processes offers:

- Multiple channels for clients to use in contacting an agency.
- A range of assessment tools to determine the complex needs of clients, the many barriers that are keeping them from reaching their social economic potential, and the clients' strengths that can be utilized to overcome these barriers.
- Universal program screening to determine the best possible combination of programs that will be effective (both in cost and outcome) in meeting client needs and overcoming these barriers (both in cost and outcome).
- Client-centered tools that make it as easy as possible for clients to get the care and protection they need.

## 2.2 Needs to Delivery - Cúram Integrated Case Management

Cúram Integrated Case Management includes a set of business processes that seek to assess client need and deliver programs in the form of benefits and services.

These processes follow a logical progression from assessing client needs through to delivering programs and services. The initial contact between clients and agencies can happen in numerous ways including triage, screening, and intake. Working with the client, the agency assesses the client's complex needs and matches those needs to a range of potential programs and services. Real world data is captured about the clients and is used to determine their eligibility and entitlement for benefits which are then delivered to the client. In addition to benefit eligibility, services may be recommended to meet the needs of clients.

The Cúram Integrated Case model moves away from the traditional approach that determines whether or not clients are eligible for single-benefits within set time frames and moves towards determining eligibility and entitlement across a range of benefits and services and over extended periods of time, perhaps even the client's lifetime.

It also aims to measure the effectiveness of the delivered programs and services over time. By expanding the determination period for programs and services, agencies can measure the ultimate success of these programs and services in improving the lives of its clients. Two mechanisms are provided for delivering care and protection to clients: the delivery of benefit-based programs through product delivery cases and the delivery of services through service deliveries and referrals, each of which is described in the subsections below.

## 2.2.1 Delivering Benefits through Product Deliveries

The product delivery case within the application determines eligibility for the benefits that best meet the needs of families and delivers those benefits accordingly.

The creation of a product delivery case follows on from a client's initial contact with the agency during which time the client's needs are assessed. Based on assessment results, the client is found potentially eligible for a range of benefits and one or more product delivery cases are created in order to determine real eligibility for these benefits.

Eligibility and entitlement processing results in a determination that spans the lifetime of a case (and as mentioned above, a determination can span the lifetime of a client). Within the determination are one or more decisions, each one defining the clientâ##s eligibility and entitlement for one or more components within a benefit. Financial schedules are created from any decisions that indicate eligibility and entitlement for benefits. The Cúram Financial Manager creates instructions which in turn get processed into financial instruments.

For example, a client who has recently become unemployed initiates contact with the agency. This contact leads to a screening which shows that the cli-

ent is potentially eligible for unemployment benefits for a set number of weeks since becoming unemployed. An unemployment product delivery case is created to capture more data about the client. This starts a chain of events which includes gathering evidence about the person's job loss, determining if the person is eligible for unemployment benefits, and delivering those benefits accordingly.

Over the lifetime of a product delivery case, real world data can continue to be captured, the impact of which can result in a new determination which reflects the eligibility and entitlement of client's most up-to-date information. The product delivery life cycle is described in detail in Chapter 4, *Delivering Benefits to Clients through Product Delivery Cases*.

### 2.2.2 Meeting Needs through Service Deliveries

Services can be provided directly to a client by the agency or provided by a third party provider but paid for and tracked by the agency. Additionally, client eligibility may need to be determined for some services and some services may have their own payment determination process.

To accommodate the different types of services, Cúram Integrated Case Management allows agencies to deliver services to clients using product delivery processing, Cúram Provider Management<sup>TM</sup>(CPM) processing or a combination of both. An example of a service that can be delivered by product delivery only is a Mileage Reimbursement service that reimburses a clientâ##s mileage expenses. This type of service is provided by the agency and can be used to provide payments to a client or another participant. The agency can use product delivery processing to check client eligibility based on client data and issue payments to the client or another participant in respect of the service.

If a service is provided by a third party provider, the agency can utilize service delivery processing which uses CPM financial processing to pay the provider. This type of service can be used when eligibility does not need to be determined for the service. For example, a service delivery can be used to meet a one-off need of Home Help. The agency can record details of the third party provider who provides the Home Help service, deliver the service to the client, and pay the provider in respect of the service.

If eligibility does need to be determined for a service that is provided by a third party provider, agencies can utilize some or all aspects of product delivery processing in conjunction with CPM service delivery processing. For example, if payments for a Physiotherapy service are based on a custom rate that can change over time or change based on circumstances, the agency can utilize all aspects of product delivery processing. The agency can record details of the third party provider, use product delivery processing to check client eligibility, determine the amount to pay based on the custom rate, and issue payments to the provider in respect of the service.

Alternatively, agencies may wish to only utilize the eligibility determination aspect of product delivery processing and use CPM to pay the provider. An

example of a service that can be delivered in this way is a Counselling service that is paid based on a flat rate contract. Agencies can utilize product delivery processing to determine eligibility and CPM reassessment and financial processing to pay the provider.

The process of delivering services to clients is described in detail in Section 5.2, Creating a Service for a Client. For detailed information on configuring service delivery types, see Section 3.8 in the Cúram Provider Management Guide. Note that service delivery processing works in conjunction with the Cúram Provider Management (CPM) module. That is, the services are registered as part of CPM. Therefore, the CPM module must be installed in order for agencies to deliver services to clients.

### 2.2.3 Meeting Needs through Referrals

Referrals to services are used to refer clients to a third party service provider who may have no formal contract or relationship with the agency but who may be better able to meet the client's need. Unless follow up is required, on referral, the agency usually has no further contact with a client. The process of delivering referrals to clients is described in detail in Section 5.3, *Referring a Client to a Service Provider*.

Like services, referral processing works in conjunction with the Cúram Provider Management (CPM) module. That is, the referral services are registered as part of CPM. Therefore, the CPM module must be installed in order for case workers to deliver referrals to clients.

## 2.3 Cúram Outcome Management - Assessments to Outcomes

The assessment to outcomes process allows agencies to evaluate a client and their family's entire situation in order to identify and address core needs that may not be obvious on the application but which may hinder their ability to be self sufficient.

The assessment to outcome process meets the needs of clients in two ways: through the use of assessment tools that are designed to identify client strengths and needs and through the use of outcome plans that are used to identify, schedule and manage activities that will ultimately lead to positive outcomes. For example, a family's strengths and needs can be assessed in order to identify the most appropriate activities to obtain a desired outcome, such as returning the child to the home.

An integrated case can be used to consolidate and manage a family's assessments and outcome plans. For example, an assessment can be run against multiple family members from within an integrated case in order to identify their strengths and needs.

An outcome plan is created for the family to deliver the most appropriate activities to obtain a desired, sustainable outcome. For example, service de-

liveries and referrals to services can also be added to outcome plans as activities that are recommended by the organization in order to move clients towards formally agreed objectives and sustainable outcomes.

Outcome management also includes tools that allow a family to be regularly reassessed to see if there have been any changes to its strengths or needs. Additionally tools that measure the difference that these activities make to members of the family, and ultimately how successful the outcome plan is, are provided.

For further information on how assessments and outcome management is used to meet the needs of clients, see the Cúram Outcome Management Guide.

## 2.4 The Cúram Integrated Case - A Holistic View

The Cúram integrated case provides a holistic view of a client and their family's needs as well as the steps taken to address these needs through the delivery of products, services, and ongoing case management. This includes a forum in which clients and agencies can interact with each other, a range of assessment tools to determine the best way to help clients, and a set of business processes for delivering benefits and services and creating and managing outcome plans. As such, it provides access to the full spectrum of 'needs to outcome' business processing.

Essential to helping clients achieve sustainable outcomes, the integrated case brings together all relevant people involved on a case such as product providers and family members and provides support for all those who are involved in collaborating with each other in identifying and addressing the needs of clients.

The integrated case homepage shows the family members whose needs are being assessed and addressed. It also provides a summary of the programs applied for, a to-do list that tracks the work that needs to be done by the case worker, and eligibility checks for those programs. Individual tabs within the integrated case provide access to the family members, their benefits, any timers on benefits, interviews, the family's evidence, any related cases, and a history of eligibility and entitlement results.

Essential tools are provided including shortcuts to perform key business processes aimed at assessing needs and delivering benefits and services to meet these needs.

## 2.5 Helping Case Workers Do Their Jobs - The Case Worker Workspace

The case worker workspace contains tools that assist case workers and supervisors in their daily work. Case workers frequently have high caseloads and many clients to look after and track. Therefore, it is important that they

can easily see case work that needs to be completed and can quickly access key information so that clients who need help do not fall through the cracks. The more case workers can track about their cases, the more they can ensure that clients in need are being helped.

Another benefit of the workspace is that it aids case worker efficiency and productivity. The workspace provides the following main features that help case workers to meet the needs of clients:

- Summaries of pertinent case information are provided which assist case
  workers in their daily work. For example, an appointment summary is
  available for case workers to quickly see various meeting and activity
  appointments they have for the day.
- Tasks and activities that the case worker needs to address for the day are highlighted.
- Case search and case query functionality is provided and case workers can also highlight items that they have a special interest in.
- Calendaring functionality allows case workers to schedule and track case events, activities, and meetings.

## 2.6 Achieving Sustainable Outcomes Through Ongoing Case Management

Essential to the stability of agencies is the ability to measure the success of care and protection in meeting clients needs. Achieving long-term stability for a client involves identifying the root causes of need, addressing the root causes in a number of ways, and measuring set-backs and progress along the way.

Ongoing case management tools help to ensure that clients are progressing towards self-sufficiency. Tools that provide visibility on the most up-to-date determinations, and deliveries of benefits and services are available. Tracking tools are also provided to check on payment corrections, issues raised on a case, and ongoing legal proceedings. Case workers can manage the progression of their product delivery cases during ongoing case management. This includes tools to manage the most-up-to-date information about a client and the ability to use this data to re-evaluate those needs.

## 2.7 Supporting Agency Operations Through Configuration

Many aspects of Cúram Integrated Case Management are configurable. The purpose of this is to enable agencies to decide how they would like the application to behave with minimal disruption to its day-to-day operations. Increased configurability significantly reduces, and often eliminates, the need

for agencies to customize source code by allowing them to configure at runtime the case information that is displayed in the application as well as how certain case processing operations are handled by the system. The following is a summary of the main aspects of the Cúram ICM model that are configurable:

- How and what case information is displayed throughout the application is configurable. This includes the ability to configure integrated and product delivery cases, as well as choose which case types are displayed in case listings and the order in which evidence is displayed on the evidence dashboard. Agencies can customize the appearance of the application user interface to highlight key information to case workers. The purpose of these options is to enable case workers to easily access the information they frequently use.
- The agency can configure the types of cases that are displayed as filter options when case workers access and search for case information. The agency can also determine the type of search facility it wants to use. Two types of case search are also supported: standard and enhanced.
- The agency can dynamically create new products, evidence for capturing real world data about clients, and the rules that are applied to this evidence in order to determine eligibility and entitlement for the new products.
- Cúram Express Rules (CER) is a rules language aimed at providing a clear expression of business rules. As part of defining rules, agencies can configure categories of rules that determine how eligibility results are displayed and also configure explanations of eligibility results that aid case workers and clients understanding of eligibility and entitlement decisions made by the system.
- CER is key to enhancing and encouraging collaboration. Documenting and displaying eligibility, entitlement and enrollment business rules in a human readable format will allow for greater transparency to the client and will aid client advocacy groups in explaining and assisting clients with the eligibility and enrollment process.
- Rules Editor functionality allows business and administrator users to define new rules and scripts together. Views that allow administrators to easily change rules and see the changes clearly on the system are provided which result in less errors being made and more accurate eligibility results for clients.
- The agency can fully configure services and service referrals to meet its requirements. Aspects of service delivery and referrals to services processes that can be configured include how the provider of the service is specified, whether a service or referral is applicable to one or multiple clients, and the strategy for processing and issuing payments in respect of services to clients and providers.
- Agencies can configure various aspects of system behavior as cases are

moved through different stages in their lifecycles. Configurable aspects of system behavior include how the system determines the initial owner of a case, case closure processing, and case reassessment processing.

## Chapter 3

# Assessing Needs - From Initial Contact to Integrated Case Creation

#### 3.1 Introduction

To streamline the process of getting clients the care they need, a range of triage, screening, and intake processes are available which can be used to determine the potential needs of a client on initial contact with the agency. Triage, screening, and intake typically happen before an integrated case is created for clients and their families. Each integrated case allows everything that is identified as meeting the needs of a family and all of the efforts to address these needs to be managed from one place. Typically, each integrated case is used to manage the actual determination and delivery of identified services and programs.

This chapter provides an overview of the available processes that support the agency's triage, screening, and intake requirements. This chapter also provides an overview of the triage, screening, and intake processing that happens before an integrated case is created for a client. This chapter also describes the integrated case creation process.

## 3.2 Supporting the Agency's Triage, Screening, and Intake Processes

The application provides tools that are designed to support the agency's triage, screening, and intake processes. The purpose of these tools is to establish what a person needs on initial contact with the agency and also to determine potential eligibility and actual eligibility for needs-based programs and/or need for services. The tools provided allow clients to communicate with the agency via a number of different channels and allow the agency to manage the following tasks:

- Triage: A person is asked a short set of questions to quickly identify services, screenings, and products which may meet their needs.
- Eligibility Screening: Individuals are screened to determine their potential eligibility for specific programs offered by the agency. If eligible, these programs are typically delivered to the individual in the form of benefits.
- Program Intake Processes: A person may be eligible for needs-based and/or protection-based programs identified as a result of screening or independent of the screening process.
- Life Event Management: Life event management allows the agency to manage the reporting of a person's life events, such as the birth of a child, marriage, or a change of address or employment.

The following infrastructures are provided that include tools and functionality that enable the client and/or agency to perform intake: Cúram Universal Access<sup>TM</sup> (UA) and Cúram Common Intake (CCI). UA is a citizen focused module that allows clients themselves to interact with the agency online using a web self-service application. For more information on UA, see the Cúram Universal Access Guide.

CCI provides a set of features and functionality that an agency can tailor to meet the requirements of its internal intake processes. For more information on CCI, see the Cúram Common Intake Guide.

## 3.3 Triaging Clients for Programs and Services

Triage is used to quickly identify and recommend services, government programs, and screenings which may meet a person's needs. The aim of triage is to quickly recommend potential courses of action for a person who contacts the agency for help. Typically triage is performed when a person has had no prior interaction with the agency and is completely unaware of the help that may be available to him or her. For example, a person may have an emergency need for food for his family and might be unaware of benefits that are available to meet this need.

During the triage process, the person is asked a short set of basic questions related to the following core needs: food, shelter, health, financial, education, and safety. Based on the person's answers to these questions, the system recommends services and programs that may address those needs. The person can be referred to various service providers identified by the system, continue to screen for potential eligibility for recommended programs, or apply for programs directly.

## 3.4 Screening Clients for Programs

Screening is used to establish a person's potential eligibility for particular programs. For example, a person may be potentially eligible for Food

Stamps benefits based on his or her answers to the screening questions. The difference between triage and screening is that triage directs a person to screen for programs, it does not identify if the person is potentially eligible. Screening determines a person's potential eligibility for one or more programs. Examples of screenings include Cash Assistance and Food Stamps. A person may choose to be screened for eligibility after completing the agency's triage process or may choose to screen independently of triage.

During the screening process, a person is asked more complex questions about their life and their answers are recorded and stored on the system. For example, personal details such as the person's name, date of birth, address, living arrangements, and family relationships are recorded. The answers recorded are used to establish if the person is potentially eligible for benefit-based programs and if he or she has already been registered on the system as a client of the agency. Family relationships that are recorded are also automatically stored on the system.

Depending on the type of screening, answers to more complex questions may also be recorded. For example, some questions may relate to the client's resources, income, or household circumstances. When all of the screening questions have been answered and the screening is completed, the system determines the programs for which the person is potentially eligible by running screening rules against the answers to the various questions.

Based on the screening results, the client can submit an application for the programs identified. Alternatively clients can apply for programs independent of the screening process.

## 3.5 Program Intake Processes

In general, clients may be eligible for needs-based programs or protection-based programs. Examples of needs-based programs include Cash Assistance and Unemployment Benefits. An example of a protection-based program is "Child Welfare". The intake processes that are used by agencies can differ greatly depending on whether the intake is needs-based or protection-based.

For needs-based programs, a client is usually required to fill out an application form, and the agency determines eligibility for programs based on the information provided in the application.

Basic information required to submit the application for benefits, such as client name and address, is then captured. The agency determines the client's actual eligibility for benefits based on the information provided in the application. A person can apply for benefits using a number of methods which include:

- submitting an online application to the agency
- filling out an editable application form
- calling to the agency in person

#### over the telephone with a case worker

During the application process, the person provides the agency with detailed information on personal details, income, expenses, employment, education etc. This information is recorded as evidence on the client's case by the agency when the case is created and is used by the system to determine eligibility for benefits. The application provides a needs-based program intake as part of Cúram Common Intake (CCI). See the Cúram Common Intake Guide for more information.

Intake for a protection-based program such as "Child Welfare" can include the capture, via a number of channels, of a reported incident of suspected child abuse or general concerns regarding a family situation. All relevant information such as allegations, are captured so that the claims can be investigated by the agency. The application provides a protection-based program intake as part of Cúram for Child Welfare (CCW). See the Cúram for Child Welfare Guide for more information.

## 3.6 Creating an Integrated Case

If the agency has a defined intake process, an integrated case can be created for clients automatically on program application. Alternatively, if no intake process is defined by the agency, a standalone integrated case can be created.

Each integrated case is initially created for a person or a prospect person to manage the delivery of benefits, services, and activities identified to meet his or her needs. On integrated case creation, a product delivery case is created to deliver benefits to clients. Note that if the agency has a defined intake process, the product delivery case is created automatically when a benefit is authorized on the application.

Service deliveries and referrals to services can also be added to an integrated case to deliver appropriate services to clients. The process for delivering benefits to clients through product delivery cases is described in detail in Chapter 4, *Delivering Benefits to Clients through Product Delivery Cases*. The process for delivering services deliveries and referrals to services are described in detail in Chapter 5, *Delivering Services to Clients through Service Delivery and Referrals*.

An integrated case is created for a person or a prospect person who is already registered on the system or the caseworker has the option of registering the individual as a person or prospect person when he or she creates the case. Note that if the agency is utilizing intake processing, the individual is registered as part of the intake process. For information on person and prospect registration, see the Cúram Participant Guide. During manual creation of an integrated case, the integrated case type is selected.

An integrated case has three steps in its lifecycle: case creation, case closure, and case reopen. When an integrated case is created or reopened, its status will be set to 'open'.

An example of when an integrated case may be closed is where all of the associated benefits and services have ended. If the clients on the integrated case re-apply for benefits/services at a later date, the integrated case may be reopened to resume the delivery of benefits and services without the need to create a new integrated case. Alternatively, organizations may choose to leave integrated cases in an open state regardless of the status of the associated benefits/services.

### 3.6.1 Assigning Initial Case Ownership

Cases are created and managed by case workers with case owner and case supervisor user privileges.

The case owner is responsible for a case. Typically, the case owner maintains case information, gathers case information, checks eligibility, and submits the case for approval. The case supervisor verifies the data entered by the case worker and approves the case for activation.

The application provides a sample case ownership strategy which can be overridden by agencies as required. The sample case ownership strategy determines the initial case owner based on whether the case is an integrated case or a product delivery case. When an integrated case is created, the case ownership strategy automatically sets the initial case owner to be the administrator of the primary client. When a product delivery case is created, the system automatically sets the owner of the related integrated case to be the initial owner of the product delivery case. When an integrate case is reopened, the case ownership strategy sets the reopened case owner to be the administrator of the primary client.

The agency's own case ownership strategy can be configured depending on its requirements. For example, the agency may wish to direct Food Stamps cases to one set of users and Income Support cases to another set of users. For more information on configuring case ownership, see Section 8.7, *Configuring Case Ownership*.

The case owner can also be manually set to be any user or organization group such as an organization unit, position, or work queue. If the case owner is set to an organization unit, work queue or position, any users who are members of the organization group can progress work on the case. For further information, see Section 7.2, *Manually Changing the Case Owner and Case Supervisor*.

### 3.6.2 Integrated Case Information

Once an integrated case is created, there are a number of levels of information that are automatically stored as part of the one integrated case: integrated case information, case participant information, and information specific to the type of case or service contained in the integrated case. Information for each of these levels can be accessed and maintained from the one integrated case.

Certain information such as notes can be maintained for integrated cases, product delivery cases, and services. These categories of information do not affect case eligibility. Other categories of information are only maintained for specific cases. For example, client translation requirements are only maintained for the integrated cases and product deliveries.

Information that is specific to product deliveries and service deliveries is kept separate from the integrated case information. For example, communications for a product delivery case will only appear on that case's list of communications. Note, however, evidence for a product delivery cases can be maintained at the integrated case level if shared across multiple product deliveries within the integrated case.

## Chapter 4

# Delivering Benefits to Clients through Product Delivery Cases

#### 4.1 Introduction

The product delivery case is used to deliver care and protection to clients in the form of benefits. This chapter provides an overview of the process of delivering benefits to clients through product delivery cases.

## 4.2 Creating a Product Delivery Case

Product delivery cases are created by default within an integrated case so that integrated case functionality can be used for that product delivery. For example, evidence can be captured at the integrated case level and shared between product deliveries for the same person or for persons within the same family.

During product delivery case creation, the case worker records who is applying for what benefit. The person applying for the benefit is called the primary client. The primary client is also the default nominee on the case. This is the person who will receive the actual benefit payments. The name of the benefit is also specified.

To aid case workers, the system allows them to view a list of any other existing integrated and product delivery cases on which the applicant is a primary client. The product delivery case details are then specified. Case details include the date that the benefit application was received by the agency and the expected outcome of the case, e.g., financial support.

Case details also include information that defines how payments in respect of the case are issued to the nominee. This information includes the preferred payment method, and the case delivery pattern. The case delivery pattern is the default payment method and frequency by which financials are issued to a nominee. For example, weekly by check. The delivery pattern in-

cludes the primary client's preferred payment method details. If the primary client has a preferred payment method, e.g., check, cash, this is displayed to indicate to the case owner the most appropriate delivery pattern.

The product delivery case is created when the case worker has recorded all of the information required. When a product delivery case is created, its status is "Open". Real world information is then captured in the form of evidence to determine the client's eligibility and entitlement.

## 4.3 Capturing Real World Information about a Client

Real world information that affects the primary client's claim for benefits is captured as case evidence. For example, in order to qualify for benefits, a client may need to provide evidence regarding his or her income. The type of evidence captured can come from different sources. For example, a client may provide a birth certificate to verify the date of birth, a university may provide a letter confirming a person's full-time student status, or a doctor may certify a person's disability.

Evidence is maintained by case workers using the evidence workspace. The evidence workspace includes the following views that allow case workers to capture, manage, and maintain evidence records easily and efficiently: the evidence dashboard view, the EvidenceFlow view, and the active and in edit evidence listings.

The evidence dashboard and EvidenceFlow views provide case workers with an entry point to pre-configured evidence types that can be captured. For more information on the EvidenceFlow view, see Section 4.3.1, *Capturing Evidence Using the EvidenceFlow View*.

Separate lists are provided for evidence verifications, issues, and for incoming evidence. For information on evidence verifications and issues, see Section 4.3.2, *Identifying Concerns with Evidence that Affect Eligibility*. Incoming evidence that is shared from other cases is only available if the Cúram Evidence Broker<sup>TM</sup> is installed. For more information, see the Cúram Evidence Broker Guide.

For detailed information on the evidence views provided and on managing case evidence, see the Cúram Temporal Evidence Guide.

### 4.3.1 Capturing Evidence Using the EvidenceFlow View

The EvidenceFlow view is an alternate view of case evidence that is available to case workers. Evidence types on a case are represented in the EvidenceFlow view as a series of ordered tiles. Each evidence type that can be recorded on the case is represented by a single tile.

The EvidenceFlow view is particularly suited to inexperienced case workers in that it allows them to quickly navigate evidence while talking to a client over the telephone.

Case workers can navigate through all of the evidence types recorded on a

case, capture new evidence, and identify outstanding information or issues that relate to the evidence type. Each evidence category configured is represented by a stack within the evidence flow view. Each stack contains the individual evidence types that can be captured. Case workers can create new stacks according to their preferences. For example, case workers may wish to create a new category of frequently captured evidence types. A case worker can do this within the EvidenceFlow view using the stack functionality.

Reminders, outstanding verifications and issues regarding an evidence type are also accessible from the EvidenceFlow view.

For more information on the EvidenceFlow view and stacks, see the Cúram Temporal Evidence Guide.

#### The Cúram Advisor

The Cúram Advisor is a tool that provides context sensitive tips such as reminders for case workers. Concerns with evidence such as issues, that affect eligibility can also be detected by the Cúram Advisor.

The Advisor is configured with rules and text that is used to detect and highlight issues with evidence. It automatically analyzes information that is recorded or known for a registered client and alerts the case worker to issues with the information that may require attention. For example, if an application is submitted by a client with missing mandatory information, the Advisor displays the issues with the mandatory information. Case workers have the option to action the issues.

For more information on configuring the Cúram Advisor, see the Cúram Advisor Configuration Guide.

## 4.3.2 Identifying Concerns with Evidence that Affect Eligibility

Outstanding information and/or concerns with evidence may affect a client's eligibility for benefits. For example, a client may be determined to be ineligible because mandatory information is missing from his benefit application or a verification by a doctor of the client's illness may be outstanding.

To help case workers identify concerns with evidence so that they can be resolved, separate lists of issues with evidence and verifications (received and outstanding) are also provided.

Note that the verifications list is only available if the Cúram Verification<sup>TM</sup> engine is installed. For more information on the Cúram Verification engine, see the Cúram Verification Guide.

## 4.4 Checking Eligibility for Active and In Edit Evidence

Evidence is captured at the integrated case level for 'open' product delivery cases before they can be approved for payment. Typically the case worker who is the case owner captures evidence and checks eligibility before submitting the case for approval. A case worker can capture evidence from the evidence dashboard, EvidenceFlow view, and active and in edit evidence listings provided. Each evidence record is assigned a status of "in edit", "active", or "canceled". in edit evidence can be modified without affecting eligibility and entitlement. When changes are applied to in edit evidence, the evidence becomes active. Active evidence is used in determining eligibility and entitlement. Therefore, the most important status for benefit eligibility determination is active as payments are only issued on active evidence records.

A case worker can choose to check eligibility for active evidence only. Otherwise the system will check eligibility on both active and in edit evidence. Eligibility can be checked on active evidence only if the case owner does not want to take in edit evidence into account. Checking eligibility with in edit evidence allows a case worker to see how the client's eligibility is affected if the evidence that is currently in edit is activated.

The check eligibility process uses the evidence to create decisions regarding the client's eligibility and entitlement over the lifetime of the case. The eligibility and entitlement rules are applied to the client's real world information to create these decisions. These decisions define whether or not the client is eligible, and if eligible, the amount the client is eligible for at a given frequency, e.g., the client is eligible for \$100 each week. For more information, on decisions, see Section 4.7, *Reading Decisions, Eligibility, Entitlement, and Explanation*.

## 4.5 Approving Payments and Activating a Case

Before a case is activated for payment, the case details and case decisions generated during the check eligibility process are typically validated to ensure that they are correct before benefit payments are issued on the case. Once the case approval process is complete, the case is activated and payments are issued to the nominee on the case.

Approval check functionality is also available that allows agencies to define the percentage of cases submitted that will automatically require approval by a supervisor. For more information, see Section 8.9, *Configuring Ongoing Case Management*.

## 4.5.1 Submitting a Case for Approval

During this stage, the case is submitted for approval. A case is submitted for approval when the case worker who is working on the case is satisfied that all case details have been entered correctly and the case is ready to progress to the payment or case activation stage. The stage has two parts: submitting the case for approval and approving or rejecting the case. The first part is when a case worker submits a case for approval. Cases submitted for ap-

proval have a status of "submitted". Eligibility is determined when a case is submitted for approval in order to provide a snapshot of the client's eligibility at that point in time.

### 4.5.2 Approving or Rejecting a Case

During this stage, the case is either approved or rejected. Typically it is the case supervisor who checks and verifies the details entered by the case worker. By approving the case, the case supervisor indicates that he or she is satisfied that the case details are correct and that the case is ready to be activated. If the case worker has approval rights as part of his or her security profile, or if the case supervisor submits the case, the case is automatically approved.

Approved cases have a status of "approved". Once a case is approved, it is ready for the next stage in the life cycle, case activation. If the case is rejected, the reason why the case has been rejected is entered and the case worker who submitted the case for approval is notified of the case rejection. If rejected, the case status returns to "open" and the case must be resubmitted to progress through the lifecycle.

### 4.5.3 Activating a Case

Cases that are approved can be activated using batch processing or manually by a case worker.

The Determine Product Delivery Eligibility batch process activates product delivery cases when the system is offline, i.e., after office hours, so that it does not interfere with normal business. This batch process determines eligibility for all product delivery cases with an 'approved' status. It then activates all eligible cases. It can also be configured to either close all ineligible cases or return the status of each ineligible case to "open". For detailed information on batch processes, see the Cúram System Configuration Guide.

Alternatively, a single product delivery case can be activated manually by a case worker. When activating the product delivery case, eligibility is determined and the case is activated for payment provided it is eligible. If ineligible, and the system has been configured to close all ineligible cases, the product delivery case is closed. Manual activation provides support for front office payments, e.g., to provide emergency assistance to persons in need. Once a product delivery case is 'active', benefits for that case can be delivered.

## 4.6 Applying Rules to Real World Information to Make Decisions

Agencies use rules to apply legislation to a client's real life circumstances in order to make decisions regarding the client's eligibility and entitlement for

care and protection programs. Typically a person must satisfy rules in order to qualify for benefits. For example, to be eligible for income support, a person's income must fall below an income threshold. Rules can also be used to make connections between different pieces of information, for example, to determine the members of a household based on sharing a common address.

Real world information is captured for clients as case evidence in order to determine their eligibility and entitlement. For example, when a person's income amount is recorded and eligibility is checked, a pre-defined eligibility and entitlement rule set that defines the income threshold amount that must be satisfied is applied to the income amount recorded. Based on these results and the results of other rules governing income assistance eligibility, the system determines the person's eligibility and calculates his or her entitlement.

When eligibility is checked, a determination is created that contains a set of decisions that are created over the lifetime of the case, each decision including an effective period of time, eligibility result and an entitlement amount and frequency (if eligible).

In addition to eligibility and entitlement results, the application allows agencies to provide decision details explaining decision results, and also provides the ability to display the key decision factors that were involved in determining eligibility such as a change to total household income. Any changes to the real world information referenced by the eligibility and entitlement rules will automatically result in the rules being re-run against the new information. This can result in a new set of eligibility and entitlement decisions over the lifetime of the case. This can also result in a new set of key decision factors and new decision details. For more information on decisions, key decision factors, and decision details, see Section 4.7, *Reading Decisions, Eligibility, Entitlement, and Explanation*.

## 4.7 Reading Decisions, Eligibility, Entitlement, and Explanation

When a case is approved, activated, reassessed and/or eligibility is checked, eligibility decisions are generated based on the application of pre-defined rules to a client's real-world information. See Section 4.4, *Checking Eligibility for Active and In Edit Evidence* for information on checking eligibility. Each decision generated indicates whether a person is eligible or ineligible and displays the person's benefit entitlement over a period of time. Each benefit entitlement consists of the financial and/or non financial results generated when eligibility is checked.

To aid case workers in understanding the eligibility results, the application provides a number of decision views. Each view gives a case worker a different level of information that describes how eligibility and entitlement decisions were reached.

The main purpose of decision views is to provide clarity on why a person is

or is not entitled to benefits and to help case workers to understand how the system arrived at that eligibility result. For example, a client may be deemed ineligible for benefits, because he or she mixed up day and month on his or her benefit application. Decision views provide a level of clarity to the process of eligibility and entitlement determination. Providing case workers with visibility on why the person is or is not entitled reduces the likelihood of benefit fraud and also allows case workers to give clients a clear picture of their eligibility and entitlement results thus enabling case workers to do their jobs more effectively.

Each decision includes the client's total entitlement amount. The agency can configure in what frequency the entitlement amount is to be displayed, i.e., monthly, daily, or weekly. For example, the agency may wish to display entitlement amounts in a weekly format of \$70 per week instead of a daily format of \$10 per day. The frequency at which the entitlement amount is displayed is defined when a product is configured during administration.

A number of views are provided that allow case workers to understand the decisions made over the lifetime of the case. For example, a graphical view includes key decision factors that have influenced eligibility decisions. Key decision factors help case workers to understand how a client's real-world information impacts their eligibility and entitlement by providing an understandable explanation of the determination results. For example, if a client receiving single mother benefits reports a significant life event such as a marriage, that client may no longer be eligible to receive single mother benefits.

The following are examples of key decision factors: "Susan Smith got married", "head of the household moved out", "Susan Smith turned 19". This allows the agency to determine what a rule means in the real world and provide an explanation of the rule that the case worker and client can both understand. For information on configuring key decision factors, see Section 4.7.1, *Viewing Decisions over the Lifetime of a Case*.

The decision details view of an individual decision provides a user-friendly explanation of how the decision was reached. The information displayed to the case worker can be grouped into categories and the information displayed on a number of different tabs e.g., a Household Information tab might display detailed information on the composition of the household and a Medical Expense tab detailed information on the family's medical expenses. The agency can configure the categories of information and the order that the decision information on each tab is displayed as part of the application administration.

Note that the agency must develop the rules used for determining eligibility and entitlement, displaying decisions, and displaying explanations. For more information, see Section 8.5, *Defining Rules and Assigning Them to Products*.

## 4.7.1 Viewing Decisions over the Lifetime of a Case

When eligibility decisions are created at various points in the lifecycle of a case, a determination is created that groups these decisions together, allowing a case worker to clearly see what decisions are associated with a particular determination of client eligibility. The following views are provided to allow case workers to view determinations and decisions over the lifetime of a case: the current determination view, the determination history view, the graphical determination view, and the decision details view. Each of these views is described in the sections below.

#### Current Determination and Determination History Views

The current determination view displays the list of decisions for the most recent determination on the case resulting from initial case assessment or case reassessment on an active case, e.g., the determination created during activation of the case or upon subsequent reassessment of the client's eligibility due to a change in circumstance. Within the determination, the decisions are listed over the lifetime of the case. Each decision includes the coverage period for which the decision is effective, the eligibility decision, i.e., eligible or not eligible, and the entitlement amount. The entitlement amount is the amount that the client is expected to be paid for the period in which the decision is effective. The format of the entitlement amount is configured during administration.

The determination history view displays a list of all the determinations on an active case that resulted from initial case assessment or case reassessment. One current determination is displayed that includes the date and time the determination was made, the reason the client's eligibility was determined, and the name of the user who made the determination. All other determinations displayed in the history previous determinations that have been superseded by the current determination. A determination may be superseded as a result of case reassessment. For example, a client who was previously ineligible for income support benefits may be found to be eligible due to the loss of his job and the recording of new evidence that places his current income below the income threshold. On reassessment, a new eligibility determination is displayed for the current date which contains the client's eligibility decision. This determination supersedes the previous determination. From both the current determination and determination history view, the case worker can view a summary of the decision details for each decision in the determination.

#### Graphical and List Determination View

The graphical view of a determination is designed to allow case workers to clearly see the connection between the eligibility result and the real world information that influences it. This level of visibility highlights the real information that is affecting eligibility and entitlement and plots it across a timeline that spans the lifetime of the case.

For example, if the system determines that a person is ineligible from the case start date because his or her income was \$300 and that person's income

drops below the required threshold to \$200 a week, the system may determine the person to be eligible, and a key decision factor is displayed on the date of the change in income, along with a new decision for the change in eligibility that explains that the person is now eligible because his or her income information has changed.

Allowing the agency to see how the update to the real world information has affected the person's eligibility result empowers case workers to explain to clients how their circumstances have affected their claims. For example, a client seeming to meet all eligibility requirements for a care and protection program, but who forgot to prove their citizenship, may contact the agency to enquire about his or her eligibility status. By displaying citizenship verification status as a key decision factor, the case worker can quickly observe the reason for the client's ineligibility status and communicate this to the client.

The graphical view can also be displayed in a flat list view. This view includes all decisions and key decision factors displayed in a list format in chronological order.

#### **Decision Details View**

The decision details view displays detailed information about an individual decision within a determination and is accessed by selecting an individual decision from within the list of decisions of a determination. This view consists of multiple tabs that each display information to assist the case worker in understanding the client's eligibility. For example, a person may be eligible for an income assistance benefit and also a medical expense allowance.

The total amount of entitlement and the household members included in eligibility determination might be displayed on a Summary tab, while another Medical Expense tab might display detailed information on the family's medical expenses that were factored into determining the medical allowance. The decision details view can also be used to display information on why a person was found ineligible.

Cúram ICM provides support for creating decision details views. Decision details rules are used to determine what information is displayed in this view and are configured during administration. For more information on decision details rules, see Section 8.5.3, *Assigning Rules to Products*. Display categories and decision details rules which include a display order that the system automatically uses to determine the order in which the tabs will appear can be associated with a product. For information on configuring products, see Section 8.3, *Configuring Products*.

## 4.7.2 Comparing Decisions

Decision comparison functionality is provided which allows case workers to compare a decision within a determination to a previous decision within the determination. Decision comparison provides the ability for a case worker to easily understand why client eligibility and entitlement has changed. For example, a case worker may wish to use the decision comparison to better understand if income was a primary factor in the change to a client's eligibility.

Case workers can also use the decision comparison functionality to understand why an over or under payment was generated by the system. The decision comparison view allows the case worker to see the previous and current values of key information used in determining why the client's eligibility has changed, for example, a change in the client's net countable income may have caused the client to become ineligible for benefits.

Note that the agency must develop the decision details rules used for displaying decision comparison information. For more information, see Section 8.5, *Defining Rules and Assigning Them to Products*.

## 4.8 Scheduling Payments and Bills- the Financial Scheduler

Payments and bills are automatically scheduled when a product delivery case is eligible and active. Payments and bills are scheduled, processed, and issued to the primary client or to a case participant who the primary client has nominated as a recipient for a case component. The following subsections provide an overview of the financial processing that occurs when a product delivery case is activated.

For detailed information on financial processing, see the Cúram Financials Guide.

## 4.8.1 Case Nominees and Case Components

A case nominee is a case participant who receives payments or bills. A case nominee is assigned to each case component. Case components for product delivery cases are units of entitlement, typically financial units of entitlement, that are available with a benefit. For example, personal component, child dependent, spousal dependent are units of financial entitlement with a benefit case. A personal component could be offered at a weekly rate of \$70. If the product delivery case is found eligible for this case component, the component will be delivered to the client who was nominated to receive this case component. So in this example the nominee may receive \$70 payment a week.

A case may have more than one case component associated with it. A nominee can be assigned to one or more of these components. Because there are multiple case components, there can also be multiple case nominees. As such, a component assignment history is maintained for each case component. This keeps track of all the nominees assigned to one component over time.

The default case nominee for all case components is the primary client. A

case participant other than the primary client can be nominated to receive benefits for a case component. For example, a person's employer can be nominated to receive a person's benefits. Alternatively, a case participant can be specified as the default nominee. This automatically assigns all case components to the case participant and eliminates the need to assign each component separately. Each time a case component is assigned to a nominee, a delivery pattern from the list of nominee's delivery patterns and a component assignment date must be entered.

### 4.8.2 Scheduling Payments and Bills

Financial components are used to schedule payments and bills. These financial components define the amount, delivery method, frequency, and period for which a payment or bill is issued to a nominee. Each financial component is used to deliver eligible case components to case nominees. When a product delivery case is activated, these financial components are created.

Financial components are delivered to nominees according to the nominee's delivery pattern. Delivery patterns are used to create financial schedules for each case component delivered. The delivery pattern includes the method and frequency with which the benefit should be delivered to a nominee. For example, a nominee's delivery pattern might indicate that all benefits should be delivered to the nominee by check at the start of every month.

For example, a nominee receives a max personal case component with a weekly rate of \$70. According to the nominee's delivery pattern, the nominee receives payments in check form at the start of every month. Based on this delivery pattern and the case component, a financial component is created for the max personal case component. This financial component will be issued by check to the nominee at the start of every month in the amount of \$280 (at a rate of \$70 per week for the month).

Delivery pattern information is set up for each case nominee. A new delivery pattern can be specified for a nominee as part of nominee component assignment or the default delivery pattern specified for the case is used. Changing delivery patterns for a nominee enables agencies to pay different case components to the same person at varying frequencies. Alternatively, a case worker can create a new nominee and assign one or more case components to the nominee or set a nominee to be the default nominee on the case to change the component assignment to suit the client's needs. A nominee can also have multiple active delivery patterns for the same period of time.

Depending on how the product delivery case is configured at administration, each financial component may potentially pay indefinitely until such time as circumstances change on the case, or the end date is explicitly set on the case. If the decision on an open ended case is eligible, an open ended financial component is automatically created. Financial components that are set up to pay indefinitely on a case are called open-ended financial components. This open-endedness is reflected in the decisions on the case and potentially on the resulting financial components. For information on configuring a case to be open ended, see Section 8.3.2, *Configuring Eligibility Determina*-

tion Processing.

### 4.8.3 Payment and Billing Processing

As payment and billing processing occurs, financial components are generated into instruction line items. Instruction line items are created when financial components are processed. Whereas a financial component is a schedule to pay or bill, an instruction line item is an instance of that schedule, i.e. a payment or bill. For example, a financial component indicates that John Smith is to be paid \$25 by check on January 1st. On this date, the financial component is processed into a financial instruction line item. Information carried over from the financial component includes the nominee, the delivery pattern, the amount, and the period of time the payment or bill covers. Where a financial component is a schedule, an instruction line item is the actualization of that schedule.

Financial instruction line items are rolled up into financial instructions. These financial instructions can be generated into payments or bills for nominees. For example, an instruction line item of \$100 a month is created for each month over a three month period. These instruction line items can be rolled up into one payment instruction for \$300. The \$300 is then paid into the nominee's bank account.

For more information, see the Cúram Financials Guide.

### 4.8.4 Issuing Payments and Bills

Payments and bills are issued for product delivery cases automatically during batch processing or manually by a case worker. For example, the agency can issue benefit payments automatically when a batch is run at the start of every month. Case workers can also manually issue benefit payments on cases that have been approved and activated before system batch processing is run. By manually issuing payments, case workers can issue payments to clients that are paid by methods other than EFT immediately to clients. For example, if a client is eligible from 2 weeks ago, their payment schedule may be set up to be paid weekly by cash in advance. Typically the payment would not be issued immediately, it would be generated during the scheduled financials batch run. However, as the customer is due benefit arrears, the case worker can issue a front office payment immediately instead of making the client wait until the following week for a payment that is due now.

One-off manual payments can also be issued for benefit products if the system is down and a person is in dire financial need. In this case, the case worker can issue a check manually to the person without waiting for the scheduled batch job to run. The manual payment can subsequently be recorded on the system.

### 4.8.5 Setting up Case Deductions

Case deductions are used to allocate a portion of a benefit toward a liability or a third party. Case workers can set up case deductions from benefit payments as a means of budgeting or to clear an existing debt. There are three types of deduction: applied deductions, un-applied deductions, and third party deductions.

An applied deduction is an agreed monetary amount that is subtracted from a benefit and applied toward an outstanding liability owed by the primary client or another participant. For example, \$10 of a person's benefits can be applied toward paying off an overpayment that was issued to the person.

An un-applied deduction is an agreed amount of money that is deducted from a person's benefit payment and used to make a general refund to the agency. For example, if a one-off emergency payment is issued to a person, the agency can deduct the payment from a future benefit payment issued to the person.

A third party deduction is an agreed monetary amount that is deducted from a person's benefit and paid to another participant. For example, \$15 of a person's monthly benefits can be applied toward paying off a gas bill owed to a registered utility.

For more information on case deductions, see the Cúram Deductions Guide.

## 4.8.6 Simulating Payments

Payment simulation allow case workers to preview a payment due for a future date. Payment simulation allows a case worker to see all payments and deductions due to a participant for a specific date before the payment is issued. A simulation of the payment for that week or for the delivery period of the case is displayed along with nominee details for the cover period.

The reducing balance of the total benefit payment minus any deductions per case component is also displayed. Any tax adjustments defined for the product will be simulated and applied to the benefit payment before any deductions are processed. The case worker can also save and store payment simulations. This allows the case worker to compare the effects of each simulation, along with any tax that has been initially deducted from the benefit.

The system automatically stores a list of payment simulations. Case workers can view a summary of the payment simulations that have been saved from this list. A case worker can view further detail on each simulation and can also permanently remove all existing saved simulations.

## 4.8.7 Viewing Financial Transactions

The system automatically maintains a list of all financial transactions on case. Financial transactions are the payments and bills issued for eligible and active product delivery cases. These financials are automatically included on the list of transactions as a result of payment and bill generation.

Transaction details that are displayed on the list include the amount of the processed transaction, the transaction type, e.g., payment, the name of the nominee who receives the payment or bill, the method of payment and payment due date.

Each financial transaction can be expanded to see full payment or billing details including a full breakdown of the amounts issued per eligible case component and any deductions applied to those amounts, e.g. tax deductions applied to payments. For information on financial transactions, see the Cúram Financials Guide.

## 4.9 Monitoring Recent Case Transactions

The application provides tools that help case workers to monitor their recent case transactions. Case workers can use these tools to see how their cases are progressing and identify which cases require further work.

The system automatically maintains separate lists of each case worker's recently assigned, recently approved, and recently viewed cases. These lists can be used by case workers to organize and manage their caseload. For example, case workers can use their recently assigned case list to identify new cases that are included in their caseload and plan their work accordingly.

A list of related cases is automatically maintained by the system. Case workers can also manually record case relationships between cases that are related to one another. Case transaction log functionality provides case workers with a quick summary of events that occur throughout the lifecycle of a case. A case status history is automatically maintained for each product delivery case by the system. The case status history provides an overview of the case processing statuses each product delivery case has gone though during its lifetime.

# 4.9.1 Monitoring Recently Assigned, Approved, and Viewed Cases

The system automatically maintains lists of a case worker's recently assigned, recently approved, and recently viewed cases. Each list includes both integrated cases and product delivery cases owned by the case worker. Case details displayed for each case include the case reference number, the name of the case, and the name of the primary client of the case.

The recently assigned case list displays cases that have been recently assigned to the case worker. All cases owned by the case worker or their organization unit, position, or work queue are displayed. An administrator can configure whether cases belonging to a user's organization group are displayed as part of administration.

The system displays the following additional details for each recently assigned case: the date and time on which the case was assigned to the user, the name of the previous case owner assigned to the case, and the case

status, e.g., open, approved.

The recently approved case list includes cases recently approved by the case worker as well as the cases submitted by the case worker for approval. The name of the user who submitted the approved case is also displayed.

The recently viewed case list allows case workers to see all cases they have recently viewed. In addition to general case details, the last transaction on the case, the case start date, and case status are also displayed.

Note that the length of time for which the system considers cases to be recently assigned, approved, or viewed can be set by the agency using a number of the application properties. For more information on setting these properties, see the Cúram Integrated Case Management Configuration Guide.

## 4.9.2 Viewing Related Cases

A case relationship is a link between one case and another case. Case relationships are either created manually or automatically during case processing. A relationship can be manually created between two cases for a number of reasons. For example, if a person is receiving two product deliveries fraudulently, a relationship can be created between the two product delivery cases.

Certain case relationships are created automatically during case processing. The following relationships are created automatically by the system:

#### **Integrated Case to Screening**

A case relationship is automatically created between an income support screening and an income support integrated case when evidence from a screening is transferred to an integrated case.

#### **Integrated Case to Product Delivery Case**

A case relationship is automatically created between an integrated case and a product delivery case when a product delivery case is created within an integrated case.

#### **Product Delivery Case to Product Delivery Case**

The system creates a case relationship when an over or underpayment case is created following reassessment. This ensures that there is a historical link between the over or underpayment case and the original product delivery case.

#### **Product Delivery Case to Appeal**

The system creates a case relationship record when an appeal is created to appeal a decision on a product delivery case.

## 4.9.3 Using the Case Transaction Log

The case transaction log is used to view a summary of processing that occurs within a case. A case transaction log is automatically maintained for

every case. The case transaction log is automatically updated whenever the system detects a pre-defined instance of processing for a case. For example, if a communication is sent in relation to a case, the system updates the case transaction log to this effect.

Other examples of transactions which cause the case transaction log to be updated include payment suspensions, case creation, and additions and modifications to evidence recorded on a case. In addition to these transactions, the agency can select the specific case processing functions it wants to display in the case transaction log as part of code table administration. For more information on code tables, see the Cúram System Configuration Guide.

The case transaction log displays the case transaction event type, for example, "Product Delivery Submitted" and a description of the event. Event descriptions provide the case worker with additional information about the event. The case transaction event log also displays the name of the user who performed the transaction and the transaction time. Additionally case workers can view the specific details of a transaction and view their own user details.

## 4.9.4 Understanding the Case Status History

A case status history is automatically maintained for each product delivery case throughout its lifetime. The case status history provides information on the stages the case has passed through during its lifetime. The series of stages the product delivery case passes through is called the case lifecycle.

Each product delivery case within the integrated case has it's own lifecycle. The main stages in the product delivery lifecycle are case creation, case authorization, case activation, and case closure. Other optional stages in the product delivery case lifecycle include suspending the case and subsequently reactivating the case.

Important information that is displayed for each status change includes the date and time of the status change. Case workers can use the date and time to determine the exact sequence of case processing in the event that a case passes through multiple statuses on the same date. The case status history also includes the name of the user or system process that caused the status change. Any updates to the case status history are displayed in chronological order i.e., the most recent status change is displayed first.

The following table describes the product delivery statuses in full:

	Status	Description
Open		A product delivery case status is "open" when the case is first created on the system. Performing certain processes will cause a product delivery status to return to "open" from another status. Unsuspending, reactivating, and rejecting a product delivery case will cause its status to revert to "open". If a product

Status	Description
	delivery case is ineligible following case activation, its status will also revert to "open". Note, however that this is configurable by the agency.
Submitted	A product delivery case status is "submitted" when it is submitted for approval.
Approved	A product delivery case status is "approved" when the case is approved by an authorized user, e.g., a case supervisor.
Active	A product delivery case status is "active" when the case is activated online by a system user or offline by the batch launcher.
Suspended	A product delivery case status is "suspended" when the case is suspended. Case suspension prevents the generation of future financial payments for the case.
Pending Closure	When a closure date is entered later than today's date, the system sets the status of the case to "pending closure" so it can be closed when the closure date is reached.
Closed	A product delivery case is "closed" when normal business is complete for that case. A case can be manually closed by a system user. The system will automatically close the case if the certification grace period expires or if the closure date for a case "pending closure" is reached.

Table 4.1 Product Delivery Case Type Statuses

## Chapter 5

# Delivering Services to Clients through Service Delivery and Referrals

#### 5.1 Introduction

This chapter provides an overview of the process of delivering care and protection to clients through service delivery and referrals.

The service delivery process includes creating a service, checking a client's availability to attend the service, finding the best service provider, requesting the service, and paying the client or provider. To ensure that services are successfully meeting the needs of clients, case workers can also record an outcome for a completed service and evaluate the quality of the service.

The referral process includes referring a client to a suitable provider and following up with the client if required.

Note that service deliveries and referrals processing is highly configurable. For an overview of the configuration options available for services and referrals, see Section 8.6, *Configuring Services and Referrals*.

## 5.2 Creating a Service for a Client

This section describes the process of creating a service for a client. The process includes specifying an appropriate service for one or more clients, defining the service owner, the required number of units of a service, and authorizing a provider rate.

As part of service creation, the service frequency, the client's required participation in the service, and the nominee to receive payments in respect of the service are specified. The service creation process also includes checking a client's availability to attend a scheduled service. For more information, see Section 5.4, *Checking a Client's Schedule and Availability to Attend a Service*.

## 5.2.1 Specifying an Appropriate Service for a Client

During service creation, a case worker specifies an appropriate service to meet the needs of the client. The case worker can search for a service using a taxonomy search. For information on the services taxonomy, see the Cúram Provider Management Guide. The case worker then specifies the appropriate service from the list of services returned. For example, a case worker may select a counselling service to help a client who is in need of counselling sessions.

At least one client must be specified to be the recipient of the service. If there is only one client on the integrated case, the system automatically creates the service for that client by default. Services can be appropriate for one client or for multiple clients. If a service is appropriate for multiple clients, multiple clients can avail of the same service. For example, a case worker may wish to send a married couple on a marriage guidance course.

Agencies may wish to track each service for each client separately instead of using the one service for all clients. For example, a case worker may wish to track counselling sessions for individual members of the same family separately. To allow for this, a service can be configured to be automatically created by the system for each client specified for the service. For example, if four members of a family are to receive 10 counselling sessions each, the service can be created once, and the system will automatically create four identical services, one for each client. This allows the counselling sessions for each client to be managed separately.

## 5.2.2 Assigning Service Ownership

The owner of a service is the user who is responsible for monitoring the client's participation in the service. The owner of a service can be the case worker who creates the service or any other user on the system. If a service can be owned by any user, the case worker who creates the service can assign themselves to be the owner or can manually assign ownership of the service to another user.

Depending on how the service is configured, service ownership can be automatically assigned to the case worker who creates the service, in which case, the user does not have to manually assign an owner. How service ownership is assigned is configured as part of service administration.

# 5.2.3 Recommending the Required Number of Units of a Service

Case workers can recommend that a client receive a certain number of units of a service. For example, a client may be required to receive 10 counselling sessions. In certain circumstances, it may not be appropriate for a case worker to recommend a number of units for certain services. For example, a client may be due to receive Infant Care from 1 January 2011 to 31 January

2011. In this case, the number of units of a service such as Infant Care is always 1 and is defaulted to 1 by the system. This eliminates the need for a case worker to specify the number of units.

## 5.2.4 Authorizing the Provider Rate

The provider rate is the amount of money that is paid to a provider for each unit of the service delivered to the client. For example, the rate for one counselling session may be \$50. The standard rate at which a provider must be paid is configured for each provider during service administration. This rate can be overridden by case workers. For example, a case worker may wish to pay a provider a rate above that defined in administration.

Authorizing the provider rate at the case level allows a case worker to override the rate defined for the provider in administration if required. If an authorized rate is specified by a case worker, the service must be manually approved by a supervisor.

#### 5.2.5 Recommending the Frequency of the Service

Services are usually delivered to clients at particular frequencies. For example, a client may be due to receive part time child care twice a week on a Monday and on a Thursday from 1 January 2011 to 28 February 2011. A case worker can recommend the frequency of the service. Sometimes it may not be appropriate to specify a frequency for a service. For example, a client may be due to receive 10 physiotherapy sessions between 1 January 2011 to the 28 February 2011. A client can avail of this service at any time during the period so no frequency is required.

# 5.2.6 Indicating the Client's Required Participation in the Service

Frequently, clients may be required to participate in a service so that they can keep receiving benefits. For example, in order to continue to receive cash assistance, a client may be required to participate in career guidance sessions for two hours once a week or they may have sanctions placed on their benefits. A case worker can indicate the frequency and the duration, in hours, of a client's required participation in the service.

# 5.2.7 Specifying a Nominee to Receive Payments in Respect of the Service

The designated payee for payments issued in respect of the service can be specified. For example, a client may receive physiotherapy sessions from his own physiotherapist. In this case, the agency can pay the client in respect of the physiotherapy sessions and the client then pays the physiotherapy provider. If a nominee is not specified, a default nominee will be designated according to the delivery type for the service. For information on defining

nominee functionality for a service, see Section 3.8 of the Cúram Provider Management Guide.

## 5.2.8 Specifying a Provider and Provider Type

In most cases, a case worker will specify an appropriate provider or type of provider to deliver the service to the client.

For example, an agency may wish to allow a case worker to always choose a provider to deliver a service. For example, the agency may have a contract with a particular provider and therefore it is more cost effective to utilize this provider. To cater for different scenarios, service configuration allows the agency to configure the way in which a provider is specified by a case worker at the case level. For more information, see Section 8.6.1.1, *Configuring Service Provider Selection*.

## 5.3 Referring a Client to a Service Provider

This section describes the process of referring a client to an external service provider. The referral process is used to refer a client for a service that is not paid for by the agency. Typically, the providers of these services have no formal contract with the agency. The process includes specifying a referral service, indicating to follow up with a client, and sending notification letters to the client and service provider.

Case workers can also check a client's availability to attend the service and avail of tools that enable them to find the best service provider. For more information on checking a client's schedule and availability, see Section 5.4, *Checking a Client's Schedule and Availability to Attend a Service*. For more information on finding the best service provider to meet a client's needs, see Section 5.5, *Finding the Best Service Provider to Meet a Client's Needs*.

## 5.3.1 Specifying a Referral Service

When referring a client for a service, the case worker specifies an appropriate service to refer the client for. The service specified dictates whether the referral can be created for multiple clients or for a single client. A case worker can select from a list of referral services registered in CPM or alternatively the case worker can manually specify an unregistered service.

At least one client is specified as the recipient of the service. If the integrated case has only one client, the system automatically creates the referral for that client by default. To aid case workers and clients, referral services can be appropriate for one client or for multiple clients. If a service is appropriate for multiple clients, multiple clients can avail of the same service. For example, a case worker may decide to refer a married couple for parenting classes.

## 5.3.2 Recording the Referral Date

An important part of referring a client for a service is to record the referral date. This is the expected date on which the client will meet the provider of the service. If the client does not contact the provider by the date recorded, the application provides the ability for notifications to be sent by the system to case workers after this date has elapsed to alert them to follow up with the client. For more information, see Section 5.3.3, *Specifying to Follow Up With a Client*.

## 5.3.3 Specifying to Follow Up With a Client

Case workers may wish to follow up on a referral to check if a client contacted a provider or not. To allow for this, case workers can indicate that they wish to follow up on a particular referral.

To aid case workers who have many clients to follow up with, system notifications functionality is available. If follow up is required, notifications can be automatically sent by the system to the case worker to remind him or her to follow up. If a referral date is specified, case workers can specify the number of days after the referral date after which notifications should be sent. If a referral date is not specified, the case worker can specify the number of days after the referral creation date after which notifications are to be sent.

If a case worker does not specify that a client has contacted the provider within the specified time frame, notifications can be sent to the case worker's supervisor to escalate the issue. For example, a case worker may refer a client on the 1January 2011 and specify to follow up on the referral in 30 days. An administrator specifies that notifications should be sent to a case worker's supervisor after 10 days. The case worker does not follow up on the referral. So, a case worker notification is sent on the 31 January 2011. A supervisor notification is sent 10 days later on 10 February 2011. For information on configuring notifications, see Section 8.6.2.2, *Configuring Follow Up Notifications*.

Note that referral functionality allows a case worker to record that a client contacted the provider. If the case worker specifies that the client has contacted the provider, no follow-up notifications are sent by the system.

# 5.3.4 Sending Notification Letters to the Service Provider and Client

When a case worker refers a client to a provider, both the client and the provider are notified of the referral. Clients and providers are notified using notification letters that are sent to the client and provider. The case worker can record notification text that is displayed in the provider and client referral letters. Notification letters can be issued by one of the following methods: email, by printing the letter for posting, or by printing the letter and physically handing it to a client if he or she is sitting with the case worker.

The notification letters that are sent to clients and providers are based on

pre-defined templates. For information on configuring notification templates, see Section 8.6.2.1, *Configuring Notification Templates*.

Case workers have the option of sending notification letters to the client or provider automatically or alternatively, they can send them at a later date by printing and posting them to the client or provider. If the notification letters are sent automatically, the system determines how the notification letters are issued based on the preferred communication method recorded for the participants during registration and on any communication exceptions recorded for them. If a client and/provider has a preferred communication method, such as email, the system automatically issues the notification letter by email.

Communication exceptions are methods that cannot be used for contacting the client or provider. If a communication exception of email exists for a client or provider, the system will not issue notification letters by email. In this situation and if no preferred communication exists, the system will inform the case worker so that the client and provider can be contacted by a different means, for example, by phone. For more information on how the system determines the notification method to use when notification letters are sent automatically, see Section 8.6.2.3, *Configuring a Notification Mechanism*.

# 5.4 Checking a Client's Schedule and Availability to Attend a Service

For both services and referrals, the period of time during which the service should be provided to the client is recorded. A start date must be specified. An end date is optional. Once a start date is specified, a case worker can check the client's availability for the dates specified.

Any services, referrals or meetings already scheduled for the dates specified are displayed to the case worker. This ensures that the case worker does not schedule activities during periods when the client is involved in other activities.

A configuration setting is provided for services which dictates if they should be returned when checking availability. For example, it would not be necessary to include a service that reimburses a client's mileage expenses as this is not a service that a client attends.

# 5.5 Finding the Best Service Provider to Meet a Client's Needs

If the service is provided by an external provider, a suitable service provider must be found in order to successfully deliver the service to the client. A service provider must be found for both services and referrals. When finding a service provider, it is important that the case worker considers fully the client's circumstances. In addition to cost considerations, it is equally im-

portant that services are readily available and easily accessible to clients, that they are of good quality, and that they are a good match to the client's needs.

Referral and service delivery processing allows case workers to take all these factors into account by providing functionality that helps them to find the best service provider to meet the needs of the client and ultimately assist them in achieving positive outcomes by achieving their goals.

This section describes the process of finding a suitable provider when the providers are registered on the system. The process includes locating the nearest service provider, validating provider credentials, and estimating the cost of a provider.

## 5.5.1 Locating the Nearest Service Provider

The application provides tools that allow case workers to locate the nearest service provider to a client. Finding a service provider who is accessible to the client and who also specializes in the client's needs can be a complex task for case workers. To assist them with this, they can search for a provider based on a number of criteria.

A case worker can search for a registered provider by name so that if a client requests a particular provider, the case worker can easily find them. A case worker can also find providers who are situated close to a client's address and can also filter this search by specifying to only return providers that are located within a specified proximity of the client's address.

The application also provides a specialty search that can also be used to return providers who specialize in particular areas, for example, if the client speaks Spanish, the case worker can search for all providers who are Spanish speaking.

For all providers returned from a search, a map is automatically displayed so that a case worker can easily identify their location in relation to the client's address.

## 5.5.2 Validating Provider Credentials

When deciding on providers, it is also important for a case worker to consider their credentials in order to evaluate their standard. To assist case workers, the system automatically keeps a record of important provider information. This information can be used by case workers to reach a decision about whether a particular provider is suitable to meet the needs of a client. The following information is available:

- Licenses, accreditations, other services that the provider provides, and provider service centers.
- Details of a provider's staff members can be accessed by case workers. Background check information is also available so that case workers can ensure that they do not select a provider who, for example, have staff

members with criminal convictions recorded against them.

- Incidents recorded against a provider are displayed. For example, a case worker looking for an appropriate child care provider would not select a provider that has bullying incidents recorded against it. For more information on incidents, see the Cúram Participant Guide.
- Investigations that are recorded against providers are displayed. For example, a case worker may not wish to select a provider who has, for example, a number of licensing violations recorded on the system. For more information on investigations, see Section 7.10.2, *Investigations*.

## 5.5.3 Estimating the Cost of a Service

Cost considerations are an important part of planning services for clients. To assist case workers, the system automatically estimates the cost of a service when it is scheduled for a client. Once a case worker decides on an appropriate provider, the estimated cost is updated based on the provider specific rates.

The following information, if specified by a case worker when scheduling the service, is used to calculate the estimated cost:

- Start Date of the service
- End Date of the service
- Number of units that the client is authorized to receive
- Frequency of delivery of the service to the client
- Rate that will be paid to the provider for each unit of service

The rate used in the estimated cost calculation can be one of the following:

- 1. Custom rate if a custom rate has been specified for the service this rate is used.
- 2. Authorized rate if a case worker specifies a rate to be paid to a provider this rate is used.
- 3. Contract service rate where the service authorization does not specify a service rate, the contract service rate is used.
- 4. Provider service rate where no service rate is specified for any of the above, the provider service rate is used.
- 5. Service rate where no service rate is specified for any of the above, the service rate set up as part of configuring a service offering is used. The average of the minimum and maximum amount specified is used.

Any flat-rate or utilization contracts that exist for the period specified for the service are automatically displayed. This helps a case worker to perform cost comparisons between all providers of the service.

# 5.6 Enquiring About a Service and Recording the Provider Response

Enquiries can be made to providers to request a service for a client during a specific time period. Enquiries help to prevent a case worker from scheduling a service if the provider specified cannot deliver the service during the required time frame. The provider can then respond to the enquiry and indicate if it can provide it or not.

There are three ways in which the agency can make an enquiry: by email, via the web, or over the phone. Enquiries sent by web enquiry can be accessed by the provider within their provider portal account. For information on provider portal accounts, see the Cúram Provider Management Guide. When a web enquiry is made, an email is automatically sent to the provider to alert it to the web enquiry. Enquiries made by phone can also be recorded.

Providers can respond to enquiries to indicate whether or not they can deliver the service or they can add a comment or question to the enquiry. Responses to email enquiries are sent by email and the case worker can subsequently update the enquiry on the system to reflect the provider's response or comment. Responses to web enquiries can be recorded in the provider portal. In this case, the enquiry is updated automatically by the system. An email is sent to the case worker who sent the enquiry to alert him or her that the provider has responded to the service enquiry.

The way in which an enquiry is made to a provider depends on two things: the preferred enquiry method of the provider, which can be set when registering a provider, and whether the provider has a valid email and provider portal account.

If a preferred enquiry method is specified for a provider, the method specified is the only method that can be used to send an enquiry. If a preferred enquiry method is not specified, an enquiry can only be sent by email if the provider has a valid email recorded on the system. A web enquiry can only be sent if the provider has a provider portal account. If a provider does not have a valid email recorded on the system or does not have a provider portal account, only phone enquiries can be recorded.

Statistics are also automatically provided that highlight the average length of time that it takes a provider to send an initial response to an enquiry and the percentage of enquiries that they have responded to are provided. Statistics help case workers to identify whether to contact a particular provider and the length of time they may have to wait before they get a response.

## 5.7 Approving a Service Delivery

The application provides a service delivery approval process that allows the agency to approve a service delivery before it is provided to the client. The

purpose of this is to ensure that the service details are correct. For example, a case supervisor may wish to manually approve services that are scheduled by novice users. For services which use product delivery processing to determine eligiblity, service approval triggers eligibility determination. For services which have no eligibility determination processing, service approval authorizes the client to receive the service.

Approval check functionality is also provided that allows agencies to define that a percentage of services submitted will automatically require approval by a supervisor. For more information, see Section 8.9, *Configuring Ongoing Case Management*.

## 5.7.1 Submitting a Service Delivery for Approval

A service is submitted for approval by the case worker after creation. If a service has been configured such that the specification of a provider or provider type is mandatory, the system only allows the service to be submitted once a provider or provider type has been selected.

When the case supervisor or a case worker with approval privileges submits the service, services that have no eligibility determination processing are automatically approved and the client is authorized to receive the service. For services that use product delivery processing to determine eligibility, the client's eligibility is determined when the service is submitted. If eligible, the service is approved and the client is authorized to receive the service. If the client is ineligible, the service remains in a status of "open" and the client is not authorized to receive the service.

Services that are submitted by a case worker without approval privileges will require manual approval by a user with the appropriate privileges. The system automatically assigns a task to the user with the required approval privileges. This task includes instructions to the user to either reject or approve the service. Services submitted for approval have a status of "submitted". Note that services for which an authorized rate has been specified always require manual approval by a supervisor.

## 5.7.2 Approving or Rejecting a Service Delivery

During this stage, the service is either approved or rejected. Typically it is the case supervisor who checks and verifies the service. By approving the service, the case supervisor indicates that he or she is satisfied that the service details are correct. For services that have no eligibility determination processing, once the service is approved, the client is authorized to receive the service. For services that use product delivery processing to determine eligibility, the client's eligibility is determined when the service is approved. If eligible, the client is authorized to receive the service. If the client is ineligible, the service returns to a status of "open" and the client is not authorized to receive the service.

If the service is rejected, the reason why the service has been rejected is entered and the case worker who submitted the case for approval is notified of the rejection. If rejected, the service status returns to "open" and the service must be resubmitted to progress.

Authorized services have a status of "not started" if the start date of the service is greater than today's date. Authorized services have a status of "in progress" when the start date is effective.

Note that when a service that is provided by a third party provider is approved by a case worker, an extra level of approval is provided at the CPM level which allows the agency to approve the payment of invoices submitted by the provider before the agency pays the provider in respect of the service.

## 5.7.3 Viewing Service Approval Requests

Case workers can frequently have many services awaiting approval from a supervisor. To help case workers track the number of approval requests sent to their supervisor, the system automatically maintains a list of service approval requests. When a service is submitted and subsequently approved or rejected, the approval request information is recorded. The approval request details include the name of the user submitting the service, the date of submission, the name of the user who approves or rejects the service, and the date of approval or rejection.

## 5.8 Paying Clients and Service Providers

Once a service delivery is approved, service financial processing occurs and payments are made to clients and service providers in respect of the service. This section describes how these payments are processed by the system. Case workers can also view details of the payments that are issued to clients and providers. This section also describes how the system determines how many units of a service have been delivered to a client and the actual cost of the service to the agency.

## 5.8.1 Processing Payments

Services provided directly to the client by the agency use standard product delivery processing to determine eligibility and issue payments to the client or other participant for the service. Services provided by a third party provider can be paid for in one of the following ways: using service invoices, using attendance rosters or using flat rate contracts. If payment for the service is based on receipt of a service invoice or attendance roster, the service can either use CPM processing or product delivery processing to issue the payments to the provider, depending on whether the payment amount is based on a custom rate. If payments are based on a custom rate that can change over time, product delivery processing is used to calculate the payment amount and upon receipt of an invoice or roster, issue the payments. If payments are not based on a custom rate, i.e., they are based on a fixed rate, CPM processing is used to issue the payments for the service. CPM processing is also used where payment is based on a flat rate contract.

If eligibility is determined for a service, a service authorization and one or many service authorization line items are automatically created if the client is determined eligible. For services where eligibility is not determined, the service authorization is created after service approval. Each service authorization line item represents a specific date or date range within which the client is authorized to receive the service, as well as the number of units that are authorized. The service authorization line item may also contain a specific provider or provider type from which the participant is authorized to receive the service.

Before a payment can be issued, a service authorization is required for services that are paid for based on receipt of a service invoice or attendance roster. Services where payment is based on flat rate contracts, do not require a service authorization in order to be paid. Case workers can view eligibility decisions and determinations in respect of a service for a client in the same way that they are viewed for product delivery cases. For more information on viewing eligibility decisions, see Section 4.7, *Reading Decisions, Eligibility, Entitlement, and Explanation*. For detailed information on CPM financial processing and configuring services for delivery, see the Cúram Provider Management Guide.

#### 5.8.2 Viewing Financial Transactions in Respect of a Service

The system automatically maintains a list of all financial transactions in respect of a service. Case workers can use these lists to ensure that the payment arrangements in respect of services are being met by the agency. Financials transactions can be viewed for service providers who are paid by the agency for providing services and for clients who receive payments from the agency in respect of services.

#### Viewing Payments to Service Providers

The system automatically maintains a list of all payments made to a provider in respect of invoices and attendance rosters submitted by the provider. If invoices are the means by which a provider is paid, invoice details are also displayed in addition to payments.

For services which use product delivery processing to issue payments, other financial transactions that apply to product delivery case financial processing such as deductions, components, delivery patterns, nominees and payment simulations are also displayed for the service.

Note that payments made to a provider under a flat-rate contract are not displayed. This is because, flat-rate contracts do not apply to one service. Since a flat-rate contract can be associated with multiple services, the cost of a particular service that is covered by a flat-rate contract cannot be attributed.

For detailed information on how provider payments are processed, see the Cúram Provider Management Guide.

#### Viewing Payments to Clients

A list of payments made to a client by the agency in respect of a service is automatically maintained by the system. As payments made to a client in respect of a service use product delivery processing, all financial transactions that apply to that service replicate product delivery case financial processing.

For example, the way in which the system processes deductions, components, delivery patterns, nominees, over and under payments, and payment simulations in respect of a service is the same as the way that they are processed for product delivery cases. For more information on product delivery case financial processing, see Section 4.8, *Scheduling Payments and Bills-the Financial Scheduler*.

## 5.8.3 Determining the Actual Cost of the Service

It is important that agencies are aware of the actual cost of a service. For services provided by a third party, the cost of services is automatically updated each time units of the service are delivered.

The number of units delivered is determined automatically by the system depending on the provider's method of payment. If a provider is paid based on invoices, it is updated each time an invoice is submitted in respect of the service provision, that includes a number of units delivered to a client by the provider. If a provider is paid based on attendance, it is updated each time an attendance roster is submitted, that indicates the number of units that have been delivered to the client. Case workers can review the service to identify how many units have been delivered to date and how many units remain.

In addition to determining the units of service delivered, the system automatically calculates the actual cost of the service from the payments issued in respect of the service. The actual cost of a service is automatically updated each time a payment is made to a payee in respect of the service.

## 5.9 Making Changes to a Service

A case worker may need to make changes to a service based on a client's needs. For example, a case worker may increase the recommended units of service for a client from five to ten or may change some client evidence which affects the client's eligibility and entitlement. Any changes made to a service or to the evidence used to determine eligibility for a service will be automatically reflected in the payment processing mechanism that is used by the service.

For example, any changes made to the number of units, unit amount, authorized or custom rate, start or end date of any service provided by a third party are reflected in the service authorization line items associated with the service. Any changes made to the start date or end date of any service that uses product delivery processing, are also reflected in the product delivery associated with the service.

Note that if a case worker has to cancel a service, any associated service authorization line items are automatically cancelled. If the service has an associated product delivery, that product delivery is automatically closed.

## 5.10 Completing a Service and Recording an Outcome

Service completion happens when a service has been delivered to the client. Service completion allows the agency to measure how successful the service was in meeting the needs of the client. Service completion also indicates that the service has been delivered and that the client will no longer be participating in the service.

As part of completing a service, the case worker records the outcome for the service. This helps the case worker to identify the client's progress and also to decide if the service should be used by the agency to meet client needs in the future. A completed service can have an outcome of "successful" or "unsuccessful". If a service was not successful, the reason why it was unsuccessful is also recorded.

## 5.11 Evaluating the Quality of a Service

Case workers can evaluate the quality of services delivered to a client. The purpose of service evaluation is threefold: to ensure that the service effectively meets the needs of the client, to aid other case workers in deciding on a suitable provider, and to help the agency decide whether or not to continue using a particular provider in the future. To provide flexibility to case workers, a service evaluation can either be performed before or after a service is completed.

As part of service evaluation, provider strengths and weaknesses are evaluated across a number of criteria. For example, a case worker can evaluate a provider of counselling services for punctuality, facilities, and staff. During the evaluation, a rating of 1, 2, 3, 4, or 5 is indicated by the case worker for each evaluation area, i.e. punctuality, staff. The system then automatically calculates an average rating of all evaluations submitted by case workers of the service. This average rating can be used by other case workers when deciding on a provider and in future discussions on provider quality in general.

The criteria used to evaluate a service provider is configured as part of configuring services. For more information, see Section 8.6, *Configuring Services and Referrals*.

## 5.12 Viewing the Service Delivery Status History

The series of stages the service delivery passes through during its lifetime, i.e., the progression of events for the case, is called the service delivery lifecycle. Each service delivery within the integrated case has it's own lifecycle. The system automatically maintains a status history for the service as it

passes through the various stages in its lifecycle.

Each service delivery that is created has a status which describes its progress. There are six service delivery statuses: Open, Submitted, Not Started, In Progress, Completed and Cancelled. Each status changes during service processing. The following table describes each status:

Information	Description
Open	A service status is 'Open' when first created on the system. A service also has a status of 'Open' if it has been rejected. For services where eligibility is determined, the service also has a status of 'Open' if the client is determined ineligible after the service is submitted or approved.
Submitted	A service status is 'Submitted' following submission of the service and approval is required.
Not Started	For services where eligibility is not determined, a service status is 'Not Started' when it is approved by a supervisor or if approval is not required and the start date is greater than today's date. For services where eligibility is determined, a service status is 'Not Started' if the client is determined eligible after submission or approval of the service, and the start date is greater than today's date.
In Progress	For services where eligibility is not determined, a service status is 'In Progress' when it is approved by a supervisor or if approval is not required and the start date is effective. For services where eligibility is determined, a service status is 'In Progress' if the client is determined eligible after submission or approval of the service, and the start date is effective.
Completed	A service status is 'Completed' when the service is manually completed by a user or automatically completed by the system when the last authorized unit of service is delivered.
Canceled	A service status is 'Canceled' when the service is deleted by a user. Only

Information	Description
	services with a status of 'Open, 'Submitted', and 'Not Started' can be deleted.

Table 5.1 Service Delivery Statuses

## Chapter 6

## The Case Worker Workspace

#### 6.1 Introduction

The case worker workspace is the main area where case workers manage their caseload. The case worker workspace is designed to help case workers to carry out their daily work easily and effectively. This chapter provides an overview of the main features of the workspace.

## 6.2 Accessing Caseload

The case worker workspace allows case workers to easily access their case-load. Case workers can access all cases which are directly assigned to them or to groups to which they belong. A list of cases owned by the case worker or by the organization group that the case worker is a member of is automatically maintained. The list of cases allows case workers to access the cases to which they are currently assigned as well as cases that are owned by other organisation groups that they belong to as defined by the agency during administration.

To access cases, case workers can filter the list of cases to only show cases that are currently assigned to them or by their organization group as required. The filter options that are available are dependent on the case ownership strategy set up during system administration. For example, if case ownership is assigned to users and positions, the case worker can filter the list of cases to access all cases assigned to their position within the organization structure.

An option to filter the cases further by specifying to only display cases according to their current status is also provided. For example, if a case worker only wishes to access cases which require follow-up action, the user can specify that only open, approved, suspended, and active cases be returned. The case statuses displayed as filter options reflect the case statuses that the case can have.

The filter criteria specified by the user is automatically stored so that the case worker does not have to enter it again the next time he or she returns to the page. The user can change the display criteria as required.

## 6.3 Viewing Daily Workload Summaries

To assist case workers in their daily work, the case worker workspace allows case workers and their supervisors to view summaries of their daily workload. When a case worker logs into the application, a number of predefined summaries are displayed. For example, if defined, a summary of currently assigned cases which may require attention are displayed to the case worker. These allow case workers to see at a glance cases and activities that relate to them and need to be addressed. Each summary provides a snapshot of work for the day and allows for quick access to the information. Case workers can access the full list view for each summary in the relevant area of the workspace and complete the relevant actions on particular information from there.

Case workers and supervisors have the option to change the display of the summaries from the case worker workspace if required. For example, a case worker who rarely uses the case queries summary can turn off the display of the My Case Queries summary if required.

The following summaries are available to case workers:

#### **Quick Links**

Quick Links allow for quick access to common actions. Examples of common actions include searching for a case, searching for a person, searching across all participants, and changing the application login password. The operations displayed are configured as part of the application administration.

#### My Appointments

The My Appointments summary displays appointments recorded in a case worker's calendar. Case workers can use the summary to view activities scheduled for today and next week. Any activities scheduled for today are highlighted to the case worker. Case workers can complete the relevant activities within their case calendar. For information on the case calendar, see Section 6.4 below.

#### **My Tasks**

My Tasks summary lists the work that is available to the case worker. Each task is listed in order of priority.

#### **My Items of Interest**

The My Items of Interest summary lists items of interest recorded by the case worker. For example a case worker record a case that he or she is tracking as an item of interest. The case worker can access each individual item of interest and access his or her list of items from the pod.

#### **My Case Queries**

The My Case Queries summary displays a list of the case worker's recorded case queries. For example a case worker may run a query on a case that was previously assigned to him or her. A case worker can access recorded queries from this list and run new queries from the summary as required.

#### **Caseload Summary**

The Caseload summary provides a graphical representation of a case worker's open, submitted, suspended, active, approved cases. The graphical view allows case workers to easily track their workload. This axis displayed in the graph is automatically updated according to the number of open, submitted, suspended, active, approved cases. Case workers can also access their case load from the summary

In addition to the summaries provided for case workers, the following summaries are provided specifically for supervisors to track their daily work and the work of their team.

#### **My Work Queues**

A supervisor can use this summary to quickly see his or her work queue.

#### **Assigned Workload**

A supervisor can use this summary to quickly see a snapshot of the assigned workload of the case workers managed by the supervisor.

#### **Open Workload**

A supervisor can use this summary to quickly see a snapshot of the open workload of the case workers managed by the supervisor. This is the workload that has not been assigned to individual case workers.

#### **My Organization Units**

A supervisor can use the organization units summary view details of his or her organization units.

#### 6.4 The Case Calendar

A case calendar is provided to help case workers and clients to schedule and maintain events, activities and meetings that relate to the agency's cases and clients. In particular, the case worker calendar is used to schedule activities and meeting that address the needs of the agency's clients and their families. The case worker calendar provides a view of a client's activities and meetings and also provides access to calendaring information about a client's cases and family members. Calendar activities and meetings can be set monthly, weekly and daily.

A calendar is provided for each integrated case and product delivery case so that associated case events can be displayed in a case worker's calendar. The integrated case calendar is maintained at the integrated case level and is used to track events or activities automatically created by the system as a result of case processing; or manually created by a case worker. Integrated

case calendars are also used to schedule meetings for case workers or participants concerning a case or client. Product delivery calendars are used to track case events and meetings created at the product delivery case level.

For information on using the case calendar to track and schedule events, activities, and meetings, see Section 7.5, *Using The Case Calendar to Schedule Meetings and Track Case Events*.

For more information on the different calendars provided by the application, see the Cúram Calendaring Guide.

## 6.5 Searching and Querying Cases

The case worker workspace provides features that allow case workers to perform case searches and query cases. Case searches can be performed at the organization level and also at the user level. The user level case search functionality provides access to any cases assigned to the case worker who is currently logged in. The organization level case search accesses any cases existing within the organization regardless of the case worker they are assigned to.

#### 6.5.1 Searching for Specific Case Information

Case search functionality is provided for accessing specific case information across the whole organization. The application provides two types of case search: standard search and enhanced search. A quick search facility is also provided which allows the case worker to perform a quick search by reference number.

The standard case search allows case workers to search for a case by case reference number and/or client reference number. The client reference number is an alternative ID for a person or prospect person, for example, a passport number or medical card number. Searching by client reference number allows the case worker to search for any cases which involve the case participant who the client reference number relates to.

Case workers can also search for cases by case name, case status, or case start and end date. The case search also provides a number of filter options. These options allow case workers to filter the case search to only display cases that have an associated appeal, issue, investigation, and/or service plans. For example, if a case worker searches for benefit products with a status of open and filters the search to only display cases with issues, the search will only display open benefit products that have an associated issue.

The enhanced case search facility allows case workers to specify the same search criteria as the standard case search. The enhanced case search differs from the standard case search in that case workers can specify a client name as additional search criteria. This allows case workers to differentiate between cases for clients with the same name.

The agency can configure if a type of case is listed as a filter option on the

case search on a case-by-case basis as part of administration. For more information, see Section 8.2, *Configuring Case Types*.

The type of search facility that is used by the agency is also configured as part of the application administration using an application property. For information on configuring the case search, see the Cúram Integrated Case Management Configuration Guide.

## 6.5.2 Performing a Quick Search by Reference Number

Quick search facility allows case workers to perform a quick search by reference number from anywhere in the application. Performing a quick reference number search will search across all cases, participants, issues, and incidents. If the reference number entered matches the reference number of any existing cases on the system, the related case is displayed. Additionally, if the reference number matches an alternative ID a participant who is also associated with a case, the system displays the details of every case that the participant is associated with.

## 6.5.3 Monitoring Cases Using Case Queries

Case query functionality is provided so that case workers can monitor any cases currently or previously assigned to them. The case worker can choose specific criteria which is important to them and can then save the criteria used in the search as a personal query. This query can be run and rerun without the case worker having to specify the criteria again.

If the standard search facility is used, case workers can query cases by case reference number and/or client reference number, case category and type and case status. Case workers can filter the query further by supplying a time period or a client reference number such as a medical card number or benefit reference number. Case workers can also specify to only query cases which have investigations or cases under appeal. If the enhanced search facility is used, case workers also have the option to query cases by client name.

## 6.6 Recording Items of Interest

Case workers can record items that they have a special interest in. Items of interest include cases, issues, and participants. For example, if a case worker is working on a case that requires a high degree of tracking, he or she can record the case as an item of interest. This allows the case worker to easily and quickly access that case and eliminates the need for the case worker to search for it using the case search facility. A list of recorded items of interest is automatically maintained for each case worker. Case workers can add and remove items of interest from this list. Types of items of interest that can be marked by case workers include cases, issues, and audit plans as well as items that are not case specific such as appeals or participants.

A search facility is provided that allows case workers to search for their items of interest by type. Case workers can filter the search to display all their items of interest or a case worker can filter the item of interest search to displays only items of interest recorded for issues, audit plans, or cases.

## Chapter 7

## Ongoing Case Management

#### 7.1 Introduction

The application provides tools that facilitate the ongoing management of the agency's cases. Ongoing case management helps case workers track their cases and progress them towards achieving sustainable outcomes for their clients. Ongoing case management includes:

- Manually changing the case owner and case supervisor
- Viewing contextual information about a case
- Auditing cases
- Using the case calendar to schedule meetings and track case events
- Managing case tasks
- Ongoing client management
- Recording case notes
- Tracking issues and legal proceedings
- Reassessing case eligibility and correcting payments
- Suspending, closing, and reactivating cases

# 7.2 Manually Changing the Case Owner and Case Supervisor

The initial case owner that is determined by agency's case ownership strategy can be manually assigned to any organization object, i.e., a user, organization unit, position, or work queue by a user. This enables the organization to allow any user or users within an organization unit, position or work

queue to work on a case.

Assigning case ownership to an organization object, group, or work queue provides visibility on the case to more than one user. All users in a specified organization unit, position, or work queue will see the case in their list of assigned cases. This reduces the need to reassign a case individually if a user who is normally the case owner is on vacation or has a heavy workload. Instead, all members within the specified organization object will have equal ownership of the case and can progress work on the case.

The case supervisor can either be automatically determined by the system based on the owner of the case or explicitly set by a user. If explicitly set, the case is assigned to the specified case supervisor when it is submitted for approval. If no case supervisor has been explicitly set, the system automatically assigns the supervisor to be the user who the case owner reports to according to the current organization structure.

For more information on users, positions, organization units, and work queues, see the Cúram Organization Administration Guide.

## 7.3 Viewing Contextual Information

Contextual information is automatically displayed for cases. Contextual information ensures that when case workers navigate to different areas within the application, the main case information is always displayed. The information displayed is visually prioritized so that the case worker can quickly see key information about the case and highlighted items that may require further attention and/or follow-up action.

## 7.3.1 Viewing Contextual Information for Cases

The following contextual information is available that is common to both integrated cases and product delivery cases: the name of the primary client and photo if it exists, the case status, and case owner. The case worker can click on each photo to access client contact details if they exist. Case workers can access summary details for the case owner whether it is an organization group, position, work queue, or user.

Current special cautions that are recorded against the primary client or case members are also displayed for both product delivery and integrated case types. Special cautions are highlighted to the case worker via a special caution icon. The special caution icon is displayed beside each case member image. For more information on special cautions, see Section 7.10.5, *Special Cautions*.

Issues and legal proceedings including issues, appeals, and legal actions are also displayed if they exist. Note that appeals and legal information on issues and legal proceedings, see Section 7.10, *Tracking Issues and Legal Proceedings*. Within the case context panel for an integrated case, the number of issues associated with the integrated case and the number of appeals associated with any issues or product deliveries contained in the integrated

case is also displayed.

## 7.3.2 Viewing Contextual Information for Integrated Cases

Contextual information that is specific to an integrated case includes the members of the primary client's family that have been added to the integrated case and their relationship to the primary client. Case workers can use either the list view to view the case members in a list or they can use the photo view. When the photo view is used, each case member is represented by a photo if available and the age of the case member in years, months, or days is also displayed.

Any outstanding evidence items to be verified and in edit evidence are also displayed and highlighted to the case worker. Relationships displayed are inherited from the relationships defined for the person who is the primary client in the participant manager. If no relationship exists between the primary client and a case member, a relationship is not displayed in the case context panel for that case member. If a case member has been involved in an investigation and/or incident and has a role on the incident and/or investigation other that reporter, this is highlighted to the case worker via an icon. The case worker has the ability to roll his or her mouse over each icon to view specific details about the incident and/or investigation.

## 7.3.3 Viewing Contextual Information for Product Delivery Cases

Contextual information that is specific to a product delivery case includes details of the benefit that is being delivered, important dates such as the case start date, the date up to which the case is certified (if certification is applicable to the case), benefit payment dates, and the next case review due date.

An overview of any financials issued on the case is also displayed. If payments have been issued on the case, the last payment amount, the next payment amount, and the payment due dates are displayed. The case worker can access the individual payment details as required. If a decision is recorded on the case, the date on which the latest case decision was reached is also displayed. The case worker can access the case decision view from the date displayed to view the decision detail. This displays the current active case decision record created due to case reassessment.

If overpayments or underpayments exist on the case, these are highlighted to the case worker. The caseworker can roll over the icons to view the details. The case context panel displays the total number of overpayments that are created as a result of a reassessment of the benefit product delivery and overpayments which are not fully repaid, written off, or reversed. If underpayments exist on the case, the total number of underpayments is displayed. The caseworker can drill-down to the over and underpayment list page to view the over and/or underpayment details.

## 7.4 Auditing Cases

A case audit is an examination of the agency's cases by a third party or evaluation by the agency of a case or cases. Case audits help agencies to evaluate performance, enabling them to make decisions on ways to improve performance and affect positive changes in policy and case practice. For example, a Quality Assurance Review audit might be conducted by an agency to evaluate the effectiveness of the delivery of a Food Stamps benefit focusing on a review of the timeliness of actions and verifications completed.

For more information on auditing cases, see the Cúram Case Audits Guide.

# 7.5 Using The Case Calendar to Schedule Meetings and Track Case Events

The case calendar is used to schedule meetings and track events and activities to meet the needs of clients and their families. Meetings are appointments scheduled by a case worker concerning a client or a case to which other users or participants are invited. For example, a meeting may be held to determine the appropriate course of action for a family. An event is an individual milestone of significance to a case.

Events can be automatically created by the system as a result of case processing or manually created by a user. The following events can be created by a user manually: case activities, case referrals, case reviews, and appeals on services. An example of an event automatically created by the system is the case closure event which is created when a case is closed.

A calendar is provided for all events. Each calendar displays the name of the event and the date on which the event occurs in the appropriate date entry.

## 7.5.1 Scheduling Meetings and Recording Meeting Minutes

Meetings can be scheduled from the case calendar. Meeting minutes can also be recorded for each meeting to capture the meeting details and these can be issued to attendees and other interested parties.

The following provides a summary of the meeting functionality that is provided to enable the agency to easily co ordinate meetings:

- Multiple participants can be invited to attend a meeting.
- Meetings are integrated with tools which allow meeting requests to appear in an attendee's external email account and calendar.
- Any individual who receives a meeting invite can accept or decline the meeting invitation using his or her calendar.
- Meetings can be rescheduled.

• Meetings can be canceled.

#### **Recording Meeting Minutes**

Meeting minutes can be recorded and issued to meeting attendees and other interested parties.

The following meeting information can be captured:

- Meeting details such as the meeting subject, location, start and end/time, and name of the person organizing the meeting.
- Meeting notes and decisions.
- Meeting attendance allows the agency to capture information about meeting attendance.
- Meeting actions can be added to the meeting minutes.
- Meeting files can be attached to meeting minutes.
- Meeting summary of the meeting minute information.

Once the meeting minutes are recorded, the meeting organizer or the case worker who has recorded the meeting minutes can issue them to any person invited to the meeting.

For more information on scheduling meetings and recording meeting minutes, see the Cúram Calendaring Guide.

## 7.5.2 Scheduling Activities

Activities are events related to a case and scheduled for a specific time period. There are two activity types: the standard activity and the recurring activity. A standard activity is a once-off event related to a case and scheduled for a specific time period. A recurring activity is an activity that recurs over a period of time. An example of a standard activity is a once-off meeting between the primary client of a case and a case worker. An example of a recurring activity is a supervised weekly visit between a child and the child's parents. When a standard or a recurring activity is created, other case workers or case participants can be invited to it. Each case worker invited to the activity can accept or decline the invitation.

Activities differ from the meetings functionality described in Section 5.8.2 above in that activities are used to set up a basic event in a case worker's calendar to schedule time for interaction with participants on a case. A concerning participant such as a primary client can be specified for standard and recurring activities. Case workers invited to the activity are informed of the activity via task notification. Any time-conflicts between existing activities scheduled for a caseworker and the planned activity are automatically displayed. The caseworker can resolve or ignore any time-conflicts.

## 7.5.3 Referring Clients for Services

Referrals can be maintained for both product delivery cases and integrated cases from the case calendar. A referral is a request to a service supplier for consultation on a case. For example, a case worker may schedule a medical examination with a physician who is registered as a service supplier for a person with disabilities.

Note that this feature is only available if the Cúram Provider Management module is not installed. Cúram Provider Management provides enhanced capabilities to allow agencies to deliver services to their clients and make referrals. These enhanced capabilities are discussed in this guide. For further information, see Section 2.2.3, *Meeting Needs through Referrals*.

## 7.5.4 Setting up Case Reviews

A review is the examination of case details by another caseworker. A case review can be manually created or automatically generated by the system. The case review is conducted by a caseworker assigned to the case reviewer user role. The system ensures that the case reviewer is informed of the case review via task notification.

Automatic case reviews can be set up to occur at defined frequencies, for example, every 6 months. This frequency can be modified on a case-by-case basis.

## 7.5.5 Appealing Services

Appeals can be created from within the calendar of a product delivery case. Appeals are used to manage disputes against service providers who are providing services to the agency's clients. The service supplier who is the focus of the dispute and the appeal type is recorded. Example of appeal types include conditions for disqualification, or payment rate. Additionally, an outcome can be recorded when the appeal is complete.

Note that this feature is available if Cúram Appeals is not installed. Cúram Appeals provides enhanced capabilities to allow agencies to document a client's request to appeal decisions made within a case. For further information on Cúram Appeals, see Section 7.10, *Tracking Issues and Legal Proceedings*.

## 7.5.6 System Generated Events

Events can be automatically created by the system during specific case processing. The following table describes these events:

Event	Description
Case Submit for Approval Event	A case submit for approval event is automatically created when a product delivery case is submitted

Event	Description
	for approval.
Case Review Event	A case review event is automatically scheduled on the date a case is approved. Cases are reviewed based on the review frequency specified for the re- lated product.
Case Approval Event	A case approval event is automatically created when a product delivery case is approved.
Case Rejection Event	A case rejection event is automatically created when a product delivery case is rejected
Case Decision Event	A case decision event is automatically created on the date a decision is made on a case.
Case Payment Event	A case payment event is automatically created on the date a payment is made on a case
Case Payment Approved Event	A case payment approved event is automatically created on the date a suspended payment on a case is approved.
Suspend Case Payment Event	A suspend case payment event is automatically created on the date a payment is suspended.
Case Closure Event	A case closure event is automatically created when a case is closed.
Pending Closure Event	A pending closure event is automatically created when a case is pending closure.
Case Reactivation Event	A case reactivation event is automatically created when a case is reactivated.

Table 7.1 System Generated Events

## 7.6 Working with Case Tasks

A task is an instruction to carry out an item of work. Tasks are either manually created by a case worker or automatically created by the system. They are maintained in a case worker's workspace as part of workflow. Tasks that relate to a case can also be maintained from the case's task list. For example, a task may be created to approve a case that has been submitted for approval. This task would appear on both the case worker's inbox and on the case's list of tasks.

## 7.7 Ongoing Client Management

The application provides tools to facilitate the agency in the ongoing management of its clients, their family members and any other parties who are involved with its cases.

Ongoing client management also includes managing case members and case participants, determining the need for a translator to mediate between a client and the agency, and maintaining client contact information. Client contact information includes contracts between the agency and its case members, communications between the agency and any case participant and attachments. Typically attachments are provided to the agency by a client in support of his or her case.

## 7.7.1 Managing Case Members and Participants

In helping clients and their families achieve self-sufficiency, Cúram Integrated Case Management recognizes the importance of understanding the needs of individual family members in relation to each other. Understanding the social context of the family is essential to identifying and treating the root cause of need. To meet the needs of families, it is often necessary for the different people who provide care and protection to work together towards a common goal. To achieve this, Cúram Integrated Case Management also recognizes the importance of collaboration.

Within an integrated case, each family member is assigned the role of case member. Case members are registered persons added to an integrated case for the purpose of determining their eligibility for benefits and services. Each person who is part of the same household is automatically assigned a case member role. If relationships are recorded for the person who is the primary client of the case, case workers can add all of that person's family members to an integrated case. This eliminates the need for the case worker to add the family members one at a time. The family members that can be added are automatically derived from the relationships recorded for the person as part of participant management. For information on recording relationships, see the Cúram Participant Management Guide.

Any other individuals, organizations, or agencies that interact with the family members are assigned a case participant role. Case participants include any persons, prospects, employers, service suppliers, information providers, product providers, external parties, and utilities affiliated with the case; this includes the case members themselves.

Case participants also include any nominees, contacts, and correspondents on the case. Participant roles are created automatically based on the information entered about the case. For example, when a product delivery case is created, a primary client is added to the case's list of participant roles. The primary client is the person who the case is created for. A participant can have multiple roles on a single case. For example, if a communication is sent to a person, an additional role of 'correspondent' is assigned to the person. If the person is also the nominee who receives the benefit payment, a further role of 'nominee' is assigned to the person. In this example, the person has three participant roles on one case: primary client, correspondent, and nominee.

## 7.7.2 Determining the Need for a Translator

Occasionally, the agency may require a translator to mediate between a client and a caseworker. Translation services may be required if caseworkers working on a client's case are unable to interact with the client in his or her preferred language. A client's preferred language is recorded when the client is registered with the agency. For example, when James Smith is registered with the agency, his preferred language is recorded as "Spanish" and he cannot speak any other language. In order to interact with the client, the caseworker responsible for managing James Smith's case must be able to interact with James in Spanish or have a translator who can mediate between them.

Determining the need for a translator is evaluated on a case-by-case basis depending on the translation requirements present on an individual case. For example, a client may require translation services on one case but not on another. The need for a translator for a client can be recorded manually by a case worker or it can be determined automatically by the system. Whether the translation needs for a client are set manually by a case worker or automatically by the system is dictated by a configuration setting that is set on a case-by-case basis as part of the application administration. For more information see Section 8.2, *Configuring Case Types*.

The need for a translator is automatically determined at various points throughout the lifecycle of a case, e.g. when a new case member is added to a case, or when a case is reassigned to a new case owner. The system determines the need for a translator by checking if the case worker's language skills match client's preferred language. If they do not match, the system determined that a translator is required. A case worker may also manually update the translation requirements for a case even if they are initially determined by the system.

If a translator is required for a case participant, case workers are kept informed of it when they view the case participant details. Additionally, the system displays the preferred language of the participant who requires the translation services.

## 7.7.3 Maintaining Client Contracts

A contract acknowledges a case member's responsibilities specified under the terms of the contract. Examples of contracts include written contracts and verbal contracts.

A contract can be created for any member of an integrated case except for a member that is a prospect person. The prospect person must first be registered as a person in order for a contract to be created.



**Important** 

Contracts are maintained for integrated cases only.

## 7.7.4 Maintaining Communications

A communication is a correspondence to or from the agency. Any commu-

nication created from a communication list page within a case automatically relates to that case.

Communications can be paper, telephone, or email based. Communication functionality can be integrated with Microsoft® Word templates, XSL templates, or email servers.

The correspondent of a case communication is automatically assigned the case participant role of correspondent.

For more information on communications, see the Cúram Communications Guide.

## 7.7.5 Adding Attachments

An attachment is a supplemental file, e.g., a text document, that is attached to a case. The agency can attach scanned documents that provide information in support of a case such as a birth certificate or bank statement. Other examples of case attachments include marriage certificates, invoices, and pay slips. A range of file types are supported including Microsoft Word, Microsoft® Excel, and PDF. The system does not restrict the file size of the attachment although the agency may wish to set a limit using an environmental variable. Once the file is attached to the case, it may be accessed by other case workers who have appropriate security privileges.

Attachments can also be integrated with a content management system through the configuration of the application properties as part of administration. If an organization chooses to integrate attachments with a content management system, the file will be stored in and retrieved from the content management system rather than the application database.

For more information on how the application can be integrated with a content management system, see the Cúram Content Management Interoperability Services Integration Guide.

# 7.8 Recording Notes

Notes are used to provide additional information about a case. For example, a note may be added to a person's product delivery case stating that the person has not been attending meetings and recommending that a home visit be scheduled for the person. Note are recorded in rich text format and can be prioritized and given a sensitivity rating so that the note can only be accessed by case workers with the correct privileges. A spell check facility is also provided which allows case workers to spell check important notes. Additionally, the system can generate notes which describe case processing.

A note cannot be overwritten once it has been created on the system. When a note is modified, the system maintains a note history which includes each version of a note, the time and date the note was entered on the system and the case worker who made the note modifications. The note history also includes the reason for the note, e.g. "created as a result of case suspension".

The system may automatically generate notes at specific stages of the product delivery lifecycle. A note is automatically generated and displayed on the case notes list every time a case is suspended, unsuspended, or reactivated.

# 7.9 Certifying Case Evidence

A certification is the validation of case evidence during a set time period. For example, before issuing an illness related benefit payment, certification from a doctor may be required to validate the person's illness.

Certification ensures that only certified cases and certified periods are paid. Certifications are proposed periods of eligibility that are used to validate evidence. For example, if evidence is entered for a six month period but only three out of the six months are certified, the evidence entered for the remaining three months will not be validated until it is certified.

A default certification requirement can be set on benefits as a pre-requisite to product delivery case eligibility. If a certification is required, the case is only active during the active certification period. If there is no certification for the case, the case cannot be activated. For information on setting the default certification requirement, see Section 8.3.7, *Configuring Case Reassessment and Adjustments*.



#### **Important**

Certifications are maintained for benefit product delivery cases only.

# 7.10 Tracking Issues and Legal Proceedings

The application provides functionality that allows the agency to track issues and legal proceedings concerning case participants. Issues and legal proceedings include issues, investigations, appeals, legal actions, and special cautions. By tracking issues and legal proceedings, the agency can evaluate and if necessary, escalate, areas of concern and/or important developments that need special attention.

#### 7.10.1 Issues

Issues are created on cases to manage and resolve problems that are identified during eligibility determination. For example, if the reason given by a client for leaving a job is different to the reason provided by his or her employer, the agency can investigate this further by creating an issue on the case to manage and resolve the discrepancy.

An issue can be created for any member of an integrated case except for a member that is a prospect person. The prospect person must first be registered as a person in order for an issue to be created.

For more information on issue management, see the Cúram Issue Man-

agement Guide.

### 7.10.2 Investigations

Investigations are used to inquire into circumstances surrounding an allegation or incident. Agencies receive thousands of reports each year which must be investigated. Examples of types of allegations which require investigation include child maltreatment or benefit fraud.

For example, John Smith is in receipt of Income Assistance as he is unable to work due to a back injury. John's neighbour phones to say that John has been working on a building site for 'cash in hand' and he is aware that he is receiving Income Assistance.

Investigations allow agencies to record all relevant details concerning any type of investigation. All information surrounding the allegation is recorded including interviews, assessments, notes and communications sent or received. These provide the agency with the relevant details to complete the findings, record a decision on the investigation and ultimately resolve the allegation or incident.

For more information on Investigations, see the Cúram Investigations Guide.

### 7.10.3 Appeals

Appeals are used to appeal one or more decisions on a product delivery, assessment delivery, or prior appeal case. For example, if a person receives benefits from the agency and the payment amount is reduced, the person can appeal the decision in order to restore his or her full benefit amount. The agency itself can also appeal decisions.

#### Cúram Appeals

Cúram Appeals is an implementation of appeals processing for three types of appeals: hearing case appeals, hearing review appeals, and judicial review appeals.

A hearing case appeal is an appeal to overturn a decision on a case or a prior appeal. A hearing is held to reach a decision on a hearing case appeal. This hearing is conducted by the agency and overseen by a hearing official who was not involved in the original case.

Like a hearing case, a hearing review appeal is an appeal to overturn a decision on a case or a prior appeal. However, hearing review decisions are decided by a panel of hearing reviewers and not a hearing official. Additionally, only certain case participants can attend a hearing review appeal. A hearing review is sometimes referred to as an appeals council review.

A judicial review is an appeal that is conducted by a court to overturn a decision on a case or a prior appeal. The main difference between a judicial review and other appeal types is that judicial review appeals are conducted by

a court and as such, are outside the control of the agency.

For more information on Cúram Appeals, see the Cúram Appeals Guide.

### 7.10.4 Legal Actions

Legal actions are used to manage directives, actions or other activities concerning case participants that are conducted by a legal authority. Examples of directives and actions include hearings, petitions, orders, and requests. For example, a court may order a participant with a history of violence to stay away from the family home. Alternatively, the agency may prepare a petition for a court to detain a participant who has committed an offense. Legal actions can result from another legal action, decision or any other reason that is deemed appropriate by the agency. For example, a legal action such as a temporary custody petition may result in a temporary custody hearing that is scheduled as a result of the petition.

Three main categories of legal action are supported:

- Legal Petition
- Legal Hearing
- Legal Order

Note that the process for managing hearings for both legal action hearings and hearings held for appeals is the same.

For more information on legal actions and appeals, see the Cúram Appeals Guide.

## 7.10.5 Special Cautions

Special cautions can be recorded for case members to highlight any items requiring special attention. This information is recorded to ensure the safety of the person and the agency. Special cautions are typically directly associated with the safety of the person or the safety of others in relation to a person. Categories of special cautions include health alerts, behavioral alerts, and safety alerts.

The list of special cautions can be configured to meet the requirements of the agency. When a special caution is recorded for a case member, a special caution category and type is captured. Special caution types include suicide risk, health, such as allergies, contagious disease, special dietary needs, or safety issues, for example pertinent criminal history such as violent or sexual offender. When a special caution is no longer current, an end date is recorded which saves the special caution on a list of historical cautions. Case workers are kept informed of special cautions regarding case members via the special caution icon. When a case member has one or more active special cautions, this icon will be displayed on the person's home page. The complete list of special cautions can be accessed via the icon. Note that spe-

cial cautions can only be recorded for person participants.

## 7.11 Reassessing Case Eligibility and Entitlement

Real world information about a client is constantly changing and those changes can impact that client's eligibility and entitlement. Case reassessment updates client eligibility and entitlement as their circumstances change. Reassessment occurs automatically when evidence is added or modified that changes the result of previously created decisions. For example, when a client submits new income information, since that income has been used to determine eligibility for income support, reassessment will occur automatically, resulting in a new determination and decisions based on that income.

Reassessment can also result in the creation of a new determination when changes to evidence only impact decision details or key decision factors. For example, a child in a foster home may be moved to a new facility. This may not impact the child's eligibility and entitlement for care and protection, but is still a key factor in meeting the needs of that child and thus should be kept up-to-date in the overall representation of the explanation of the child's eligibility and entitlement.

The result of case reassessment is a new determination which includes any new decisions, new decision details information, and/or new key decision factors.

An agency can configure whether or not reassessment should occur within a closed case as part of application administration. For more information, see section 3.2.2, Eligibility Determination and Decisions in the Cúram Integrated Case Management Configuration Guide.

## 7.11.1 Over and Under Payment Processing

When a new determination is created, a comparison is done by the system to see if payments have been processed based on the previous, now superseded decision information. This comparison can result in an over or under payment. When a reassessment results in new decisions, the system automatically compares the new decisions to any existing decisions that have been processed into payments or bills. If the processed amounts are different to the eligibility and entitlement amounts in the new decisions, an over or under payment is created. An overpayment occurs when too much has been paid out on the case. An underpayment occurs when too little has been paid out on the case.

Three case types are provided allow the agency to manage and correct over or underpayments detected during case reassessment: the payment correction case type, the overpayment case type, and the underpayment case type. A payment correction case type is used to correct either an over and underpayment on a case. An overpayment case type is used to correct overpayments only. An underpayment case type is used to correct underpayments

only. The payment correction type allows case workers to see the over and under payment amounts broken down by case component. For example, a case worker can view a breakdown of an overpayment of \$400 that is made up of an overpayment of \$500 for Component A and an underpayment of \$100 for Component B.

Case workers can view information about the over or underpayments in order to understand what components were involved. Additionally, the two decisions that resulted in the over or underpayment can be compared. For more information on comparing two decisions, see Section 4.7.2, *Comparing Decisions*.

The agency itself can change the rules that get applied to a client's circumstances, e.g., legislation rates can change. Processing is provided that enables agencies to initiate the reassessment of multiple cases when this type of change occurs. It first determines how many cases will be affected by the change which allows the agency to determine the impact of these global changes. Based on the number of cases identified, agencies can decide how best to proceed. For example, if the impact is significant, case reassessments can be actioned when the systems are offline so as to minimize disruption. For more information, see Section 8.3.14, *Applying System Wide Changes to Cases*.

# 7.12 Suspending a Case

It may be necessary for the agency to suspend an "active" product delivery case. For example, the agency may suspend a person's benefits if it suspects that the person is committing benefit fraud. When a case is suspended, existing payments for the case will continue to be paid, however new payments cannot be created until the case is unsuspended.

A suspended case can be unsuspended. If, for example, the agency later discovers that the person suspected of benefit fraud is innocent. In this case, the suspension can be reversed. The status of the unsuspended case returns to "open". Future payments will only be issued after the case is approved and activated again.

# 7.13 Closing a Case

A case can be closed automatically by the system when its certification period expires or if it is found to be ineligible. A case can also be manually closed at any stage during its lifecycle or set to close on a date in the future. The case closure date, actual outcome of the case, and case closure reason is specified.

When a case is closed, all information and events associated with that case, such as case reviews and case referrals, are also closed and a case communication is printed for the primary client. A case communication is a correspondence to or from the agency that is created within a case. Additional case closure communications can be created for other case participants, such

as a service supplier on a case referral.

When a case is set to close on a date in the future, the status of that case is set to "pending closure". It will remain in this status until the closure date or until a case worker changes that closure date. If a case is already closed, the case closure date cannot be moved to a future date.

Note that by default, a case that has open tasks associated with it cannot be closed. The agency can allow cases to be closed that have open tasks associated with them by setting an application property as part of administration. For more information on this property, see the Cúram Integrated Case Management Configuration Guide.

# 7.14 Reactivating a Case

Cases with a "closed" or "pending closure" status can be reactivated. For example, a person's benefits may be discontinued once the person is in regular employment. If that person is made redundant from the new employer, and the person is once again eligible for benefits, the existing product delivery case can be reactivated.

The status of the reactivated case is "open". In order for benefits to be delivered on the case, the case must go through the case approval and case activation stages once again. Benefits will only resume after the case is approved and activated again.

# Chapter 8

# Configuring Integrated Case Management

#### 8.1 Introduction

This chapter provides an overview of the aspects of Cúram Integrated Case Management that are configured as part of administration. Case workers can create and manage integrated cases, product delivery cases, services, and referrals when this information has been configured.

Integrated cases and product deliveries are configured as case types in administration. Information that is configured for case types include information that is common to both integrated cases and products. Additionally, information that is specific to products can be configured, such as important information that is used by the system to determine client eligibility and entitlement. The application also enables the dynamic configuration of product case types, rules, and the evidence types used to determine eligibility and entitlement.

For detailed information on individual configuration settings provided, see the Cúram Integrated Case Management Configuration Guide.

## 8.2 Configuring Case Types

Integrated cases and product delivery cases are based on case types. Integrated cases are based on integrated case types. Product delivery cases are based on product case types.

This section provides an overview of the configuration information that is common to integrated cases and product to case types. This includes defining a name for each case type, configuring a case home page for each case type, and important date settings.

Additionally a setting is provided that allows the agency to create associations between related types of case. For example, the agency may wish to

group a number of related products. To meet potential translation needs of clients, a setting is also provided that enables agencies to determine whether a client's translation requirements are evaluated automatically by the system.

Important case display settings are also provided which determine where case information for the case type is displayed in the application. Agencies can use these settings to control where cases based on the case type are displayed as required.

For detailed information on the individual settings described above, see the Cúram Integrated Case Management Configuration Guide.

## 8.3 Configuring Products

Products are configured in order for case workers to create and deliver product delivery cases to address the needs of clients. A product can be a benefit that an agency provides to clients, a liability that the agency bills to clients or businesses, or a payment correction that is used to correct over and underpayments issued to clients.

Benefits are used to provide subsidies to eligible participants in the form of monetary payments. For example, a cash assistance product provides monetary payments to families in need. Liabilities are used to bill and receive payments from liable participants. In addition to liability and benefit products, payment corrections are also configured as products as part of the application administration.

A dynamic product wizard is provided that allows administrators to easily configure benefit and liability products. Each step in the wizard represents an important category of information that is configured for each product.

## 8.3.1 Configuring General Product Information

This subsection provides an overview of general product settings available that determine the basic structure of a product delivery case type.

Date settings can be configured which define the period during which product details are effective. The date settings include a start date and an end date.

The participant types that are eligible for cases based on the product can be defined. Eligible participant types that may be associated with products include person, employer, utility; information provider; service supplier; and product provider.

Product categories are used to group similar products. Examples of product categories include assistance, illness, and financial. These categories are used to define the subset of products that may be added to a given integrated case.

Case reviews can be set to occur at a defined frequencies, for example,

every 6 months. All cases based on the product will inherit the product's case review frequencies.

For detailed information on the settings described above, see the Cúram Integrated Case Management Configuration Guide.

### 8.3.2 Configuring Eligibility Determination Processing

The application allows the agency to control aspects of eligibility determination processing. Eligibility determination settings have a large impact on the determination of a client's eligibility for product deliveries and on successfully providing them to eligible clients.

To allow for situations where a client may be receiving a benefit for an unknown period of time, a setting is provided that determines if eligibility decisions created for a case can be open ended with no effective end date. A setting is also provided that determines how the system decides if a new determination result is different from the previous determination result. A setting that determines the summary information that is displayed for each decision within a determination is also provided. A setting is also provided that determines whether or not reassessment should occur within closed cases.

For detailed information on the settings described above, see the Cúram Integrated Case Management Configuration Guide.

## 8.3.3 Configuring Delivery Patterns

One or more delivery patterns can be configured for a product. These delivery patterns are available for selection at the case level when assigning a nominee's preferred delivery pattern for a particular case component. Note that delivery patterns can be added on a case by case basis. Delivery patterns define the default frequency and payment method by which financials are issued to a nominee on cases based on this product, for example, weekly by check.

Each product can be assigned a default delivery pattern which will be automatically assigned to the nominee when a case is first created. The default delivery pattern assigned to the product is provided as guidance, but does not have to be the preferred delivery pattern of the nominee. The following delivery pattern information can be defined for each product delivery pattern: maximum amount, default pattern, cover pattern, delivery method, and frequency.

For detailed information on the settings described above, see the Cúram Integrated Case Management Configuration Guide.

# 8.3.4 Configuring Product Provisions and Assigning them to Locations

The application allows agencies to configure product provisions and asso-

ciate them with locations. An example of a product provision is a benefit offered to eligible persons on behalf of the agency. The agency itself is considered a product provider if it provides products directly to its clients. Product providers are selected from product provider participants that are registered on the system as part of the Participant Manager.

Product provision settings include the start dates and end dates of the period during which the product provision is active, the estimated cost, a payment method (e.g., cash), a payment frequency (e.g., the first day of every month).

Product provisions can be delivered from one or more locations to suit the needs of the different eligible persons. For example, a training provision can be offered at different training center locations. Information recorded for the product provision locations includes the cost of the provision at the location and the period during which the provision is offered at the location. A case's product provision location can be dynamically changed to any of the provision locations stored for that case's parent product.

For detailed information on the settings described above, see the Cúram Integrated Case Management Configuration Guide.

### 8.3.5 Configuring Benefit Deduction Information

Benefit products have a number of settings that affect all deductions made from cases that are based on that benefit product. These settings apply to both third party deductions (deductions made from a participant's benefits in order to make payments to a third party such as a utility company) and standard deductions (any other deduction from a participant's benefit payment).

For detailed information on these settings, see the Cúram Integrated Case Management Configuration Guide.

## 8.3.6 Configuring Liability Information

The application provides two configuration settings that apply to liability products only. A setting that determines if over allocation is allowed on the cases based on the liability product is provided. Over allocation allows for the processing of payments of amounts of money that are greater than the amount that was billed for. For example, if an employer is billed \$100, and a payment is received from that employer for \$120, if over allocation is set up for the liability product, the full \$120 can be allocated toward the \$100 creating a \$20 over allocation.

A setting is also provided that determines whether reassessments on a liability product are either normal or reconciled. A normal reassessment is a reassessment that does not attempt to reconcile any payments that have been received against the liability amounts. Normal reassessment creates an over or underpayment based solely on changing circumstances. Reconciled reassessment creates an over or underpayment based on changing circumstances

AND related payments received and over allocation payments.

For detailed information on these settings, see the Cúram Integrated Case Management Configuration Guide.

### 8.3.7 Configuring Case Reassessment and Adjustments

Settings are provided that allow the agency to control how case reassessment processing works and how adjustments to payments are made.

Case reassessment settings are used to determine if overpayment cases, underpayment cases, or payment correction cases should be automatically created when an over or underpayment is discovered on reassessment.

Adjustment settings allow agencies to correct a client's financial account, so that the system does not attempt to generate a compensating over or underpayment. For example, adjustment functionality allows the agency to apply appropriate taxes to payments issued in respect of a benefit product. When adjustment is required for liabilities, if an employer is billed monthly for employer contributions, that employer will have one month to send payments toward the bill. If the employer does not send in payments within the month, surcharges will be applied to the outstanding bill.

Two additional case reassessment settings are provided for products that use the application Rules: the re-rate frequency and the date list. The date list is a list of dates for which eligibility and entitlement need to be established and can either be a pattern date list or an event date list.

The re-rate frequency applies to products that use a pattern date list where there is a change in circumstance. It is used by the assessment engine to compile the list of dates on which the rules engine is called in order to create case decisions.

For detailed information on these settings, see the Cúram Integrated Case Management Configuration Guide.

## 8.3.8 Configuring Certification Requirements

Certification requirement settings are used to indicate whether or not the product requires some form of certification for eligibility determination. Certification settings include a setting to specify whether or not certification is required, the frequency at which a participant provides certification, and the grace period to allow a participant to retain eligibility after certification expires.

For detailed information on these settings, see the Cúram Integrated Case Management Configuration Guide.

## 8.3.9 Configuring Bonus Payments

Bonus payments are once-off payments issued to benefit recipients. An example of a bonus payment is a back-to-school bonus given to a parent to

help alleviate the expense of new school materials. Criteria for a bonus payment is set up for product case types.

For detailed information on configuring bonus payments, see the Cúram Integrated Case Management Configuration Guide.

### 8.3.10 Configuring Absence Reasons

Absence reasons can be configured for all products. Absence reason functionality allows agencies to configure a list of absence reasons that are relevant to a particular program. For example, an agency can configure absence reasons that are acceptable for a client being absent from attending a scheduled on the job training activity to prove work requirements for receiving Income Support. An example of an acceptable absence reason is 'sickness'.

By recording absence reasons, administrators can specify whether or not a client absence from a scheduled activity is payable or deductible. An absence reason can be configured to be both payable and deductible.

For detailed information on configuring absence reasons, see the Cúram Integrated Case Management Configuration Guide.

#### 8.3.11 Setting Product Time Constraints

Product time constraints are the time limits applied to products. They relate to time limit functionality. For example, a time constraint can be added to a product defining the maximum amount of time after an eligibility decision is reached that a claimant has to appeal the decision.

For detailed information on configuring product time constraints, see the Cúram Integrated Case Management Configuration Guide.

## 8.3.12 Configuring Payment Corrections

A payment correction is an out-of-the-box product used to bill participants who have been overpaid or pay participants who have been underpaid. The agency can configure benefit and liability products such that a payment correction case be automatically created each time the system detects an over or underpayment during case reassessment.

General settings that are configured for each payment correction case includes the start date, home page, type, and review frequency. Additional details that can be configured include options related to where the payment correction cases based on this type are displayed in the application as well as a case ownership strategy. Financial details that can be specified include adjustment details, deduction details, cost, and whether over allocation is allowed.

For detailed information on configuring payment corrections, see the

Cúram Integrated Case Management Configuration Guide.

#### 8.3.13 Configuring Rate Tables

Rate tables are used to maintain values that can vary over time. An example of a rate is a Gross Income limit based on the number of people in a household. Since rate tables can be created and maintained independently, this allows for a more flexible approach to rates that apply to products. Rate tables can be used for values that are effective during set time periods. For example, the reassessment of a case might call upon values in the past, where the rates have changed considerably, and these values can be stored on a rate table.

A rate table can contain an indefinite number of rows and columns which determines the number of cells. The values of individual rates are stored in these cells. Additionally, sub-rows and sub-columns can be added to each individual row and column. A rate table can also be cloned and used as a basis for a new rate table.

Each rate table has a type and effective date. Note that if two rate tables have been created for the same type, then the effective date of the rate tables continues until midnight of the day before the effective date of the next rate table.

Rate tables are described in detail in 'Appendix A' of the Cúram Integrated Case Management Configuration Guide.

## 8.3.14 Applying System Wide Changes to Cases

Agencies are often times faced with the need to make changes to how eligibility and entitlement is determined, for example, if a rate changes as a result of a change in legislation, an administrator must change the rate using the rate table feature described above. Once this change is published, it will be taken into account when case reassessment next occurs within any product delivery case based on a product that uses the rate to determine eligibility and entitlement, for example when reassessment is manually initiated within the case or when financials are generated.

Because these types of system wide changes are likely to affect many cases, the application also provides the ability to identify and reassess all cases affected by the change. Upon publication of a change to a rule set, rate table, data configuration, or product configuration (such as a change to a product period or to the rule set associated with a product period) the system captures information about the change that has occurred.

A series of batch processes can then be executed and will automatically identify and reassess all cases that have a dependency on the change made. The batch processes will also process multiple changes together during reassessment, e.g. a change to a rate table as well as a change to a rule set.

If the result of reassessment differs from the current determination result for

the case, then a new determination will be created. For products that have been configured to not allow the reassessment of closed cases, no reassessment will occur if the case is closed.

If reassessment has already been initiated within an individual case, then that case will not be reassessed as part of the run of the batch processes.

### 8.3.15 Configuring Payment Groups

Payment groups provide the ability to group products/programs for which the payments should be rolled up during financial processing. The system will roll up the payments of products belonging to a payment group into a single payment provided the conditions for roll up are satisfied. The payments for all remaining products that are not associated with any payment group will then be rolled up into a single payment. For more information on the criteria for rolling up the payments, see section 2.5.1 Information Used to Roll Up Financial Instruction Line Items in the Cúram Financials Guide.

Creating payment groups can be useful in a scenario where the agency does not want payments from various programs to be rolled up into a single payment instruction.

For example, James Smith is eligible to receive \$100 per week from the Child Support program, \$150 per week from the Income Support program and another \$100 from the Carer's Allowance program. If no payment groups are configured, these payment instruction line items will be rolled up into one payment instruction of \$350. However, if the agency does not want the Child Support payments to be rolled up with the payments for the Carer's Allowance Program they can prevent this from occurring by creating a payment group. In this scenario, the agency can create a payment group and add the Income Support and Carer's Allowance programs to it. The system will then generate two payment instructions - one for \$100 for the Child Support program and another for \$250 which includes the rolled up payments from the Income Support and Carer's Allowance programs.

## 8.4 Configuring Case Evidence

In order to capture evidence types at the case level, each evidence type must first be configured, then activated, and associated with a case type. Associating an evidence type with a case type allows that evidence type to be captured at the case level for any cases that are based on that case type.

When an evidence type is configured to be captured on an integrated case type, that evidence can be captured on any integrated case of that type and is also accessible from the product deliveries within the integrated case.

The application allows administrators to configure both dynamic and/or non dynamic evidence types. This chapter focuses on dynamic evidence configuration. Note that dynamic evidence can be configured for integrated cases and product delivery case types only. For information on configuring non-

dynamic evidence, see the Cúram Temporal Evidence Guide.

Administrators can also design and link rules that determine eligibility and entitlement to the evidence types during rules definition and assignment. For more information, see Section 8.5, *Defining Rules and Assigning Them to Products*.

# 8.4.1 Configuring Evidence Types - The Dynamic Evidence Editor

Basic information about each evidence type is defined during administration. This information includes the name of the evidence type, its logical name, and the name of the related group of security identifiers that are generated in respect of the evidence type. Administrators can subsequently group these generated security identifiers any way they want. However if they wish to grant a user read, create, and maintain rights for an evidence type, the administrator can add the group name defined to the user's role. For information on assigning security identifiers to users, see the Cúram System Configuration Guide.

An effective from date is also specified which allows flexibility for the evidence type to change over time. For example, an agency may configure an evidence type to reflect legislation that says that says the agency must capture a participant's dental records but from the 1st of January the agency needs to capture a participant's medical records. From the 1st of January the medical record evidence type automatically supersedes the dental record evidence type. The system automatically stores a version history of the evidence type records that are effective over time and displays the correct evidence pages related to the evidence type to the case worker at runtime.

Once basic information is defined about the evidence type, a new version of the evidence type is created by default. Administrators can then modify its metadata using the Dynamic Evidence Editor. This effectively designs all of the evidence pages that relate to the evidence type. In general, the editor is used to:

- define all of the attributes that will appear on the evidence pages.
- define all of the evidence pages that relate to the evidence type.
- set up validations on each evidence type, e.g., if a case worker enters hair colour, he must also enter eye colour, or identify duplicate date ranges.
- set up parent/child evidence relationships. For example, Alien Sponsor evidence is child of Alien evidence.
- design how the evidence page will look in the user interface. For example, an administrator can drag fields on to an application page cluster, show one panel on a particular evidence page but not another.

For detailed information on configuring dynamic evidence using the Dy-

namic Evidence Editor, see the Cúram Dynamic Evidence Configuration Guide.

### 8.4.2 Activating Dynamic Evidence

Once an evidence type and its related pages are defined, the evidence type is activated in order to make it available to be associated with case types. Activating an evidence type also generates corresponding rule classes that are based on that evidence type, domain definitions, security identifiers, and evidence property files, and validates the evidence type data defined using the editor. The rules classes generated allow rules developers to refer to the evidence attributes and enables them to write eligibility and entitlement legislation rules using the CER Editor. For more information, see Section 8.5, Defining Rules and Assigning Them to Products.

### 8.4.3 Associating Evidence Types with Cases

Once an evidence type is defined and activated, that evidence type can be associated with a case. Dynamic evidence types can be associated with multiple integrated cases and products simultaneously. As part of associating an evidence type with a case, the evidence type category and sort order is specified. A quick link setting is provided which allows an administrator to specify if the evidence type and the category to which it is assigned will be available as an item in the preferred drop-down list on the evidence workspace. The preferred drop-down list allows case workers quick access to the evidence types they frequently need to access on a daily basis.

The sort order is used to determine the order in which the evidence type is listed in the temporal evidence workspace. If a sort order is not specified, the evidence type is listed alphabetically.

Evidence types are displayed in the evidence dashboard that is used by case workers to capture evidence. This allows for easy access to the evidence pages associated with an particular case type. Each evidence category can contain one or more configured evidence types.

## 8.4.4 Setting Evidence Approval Checks

Evidence approval checks are used to determine whether the evidence modified by a user can be automatically approved or whether it requires manual approval by a supervisor. The percentage specified indicates the rate of evidence changes that require manual approval, the remainder get automatically approved.

Evidence approval checks can be set up at the following four levels of application administration: organization unit, user, position, and product level. At the organization unit, position, and user levels, it must be specified whether the check should apply to a single product or all products. If no evidence approval checks exist at the user level, the system then looks for approval checks for the position to which the user belongs. If no evidence

approval checks exist at the position level, the system looks for approval checks for the organization unit to which the user belongs. If none exist here, the system checks the evidence approval checks at the product level.

### 8.4.5 Configuring the EvidenceFlow View

A number of system application properties are provided that allow an administrator to configure the EvidenceFlow view. These properties allow an administrator to dictate both the default evidence type to display in the EvidenceFlow view and the time that the system takes to reload the bottom panel of the EvidenceFlow view when a tile, i.e., a cover, within the EvidenceFlow changes. A number of other system application properties are provided that allow the administrator to dictate the interval of time that the EvidenceFlow waits when detecting changes to evidence and also to dictate whether EvidenceFlow logging and EvidenceFlow validation is enabled. For more information on these properties, see the Cúram Integrated Case Management Configuration Guide.

# 8.5 Defining Rules and Assigning Them to Products

Cúram Integrated Case Management provides a way for agencies to translate the real world scenarios that face clients into a language that can be interpreted and used to determine the benefits available to clients.

The Cúram Express Rules (CER) language can be used to meet this goal. CER aligns the business and technical representation of rules into a single expression. This means that legislation can be directly translated into rules that can be interpreted in order to make decisions regarding client needs. CER also provides a user-friendly representation of rules allowing users to understand how legislation impacts the decisions made that affect the lives of clients. For more information, see the Cúram Express Rules Reference Manual.

Legislation is complex; therefore, the rules that express this legislation are complex. In order to make the designing of rules more efficient, a range of tools are provided that assist case workers, rules designers, and product administrators in working together to dynamically design rules, link these rules to a client's real world information, and assign them to care and protection products in order to determine client eligibility. The remainder of this section describes these tools in more detail.

## 8.5.1 Dynamic Rules Design - The CER Editor

The CER Editor is a dynamic tool that is used to define and validate CER rules. The main feature of the CER Editor is the canvas in which rules expressions are designed. This canvas provides a visual representation of rules that can be understood by rules designers and business users.

New rules expressions can be created in this canvas using a menu bar that

contains common rules logic (like an 'if-then' statement determining the age limit of a child) as well as customized logic that suits agencies.

For example, the CER Editor offers quick ways in which to create rules expressions relating to household and financial units. In addition to the CER Editor canvas, short-cuts are provided to create new rules classes, to group these classes in folders, and to validate rules expressions.

For more information on how to use the CER Editor, see the Working With Cúram Express Rules guide.

### 8.5.2 Linking Rules to Evidence

In order for rules to be applied to a client's real world information, those rules must be linked to the evidence and non-evidence entities used to capture that information. Cúram Integrated Case Management provides a simple and dynamic way of linking CER rules to both evidence and non evidence based data. These links are established through the creation of a CER data configuration - xml maintained dynamically which links the rule class to an evidence type or non-evidence entity. For example, the rule class for determining income amounts would be linked to the income evidence type.

### 8.5.3 Assigning Rules to Products

Rule sets that have been created using the CER editor can be categorized and assigned to products. Rule set categories provide administrators with a facility for filtering and sorting rule sets. There are three types of rule sets that apply to the determination of client eligibility and entitlement for care and protection products:

- Eligibility and entitlement rules are normally related to or defined by legislation. These are the rules needed for the system to know whether or not a client is eligible, and if so to what the client is entitled.
- Key decision factor rules are used to provide visibility on key pieces of real world information that has changed as part of eligibility and entitlement processing. By assigning a key decision factor rule set to a product, the significant factors that were involved in determining the client's eligibility such as a change in total household income, can be displayed in the graphical representation of the determination created as a result of the eligibility and entitlement rule set associated to the product. Case workers can use this information to understand more about why an eligibility and entitlement determination turned out as it did.
- Decision details rules are used to provide a user-friendly explanation of how eligibility and entitlement was determined. These rules can be designed by an agency as required to display any information that is desirable to display to users, for example, a list of countable assets or a hyperlink to supporting information. Custom pages can be designed to display the information that is determined by these rules. Case workers can

use these decision details rules to match their understanding of the real world information to the system's determination of eligibility and entitlement.

The assignment of rules sets to products is done through the creation of a product period. The product period defines the timeframe over which rule sets are applicable for the product. An example of a product period is the 2010 calendar year.

Product periods can be open-ended, i.e., only one product period need be defined for a product, and can be applied indefinitely. When a product period is defined, the administrator can associate rules (eligibility and entitlement, key decision factor, and decision details rules) to the product period. Administrators must assign an eligibility and entitlement rule to a product period and can optionally assign key decision factor and decision details rules. As part of assigning rules to products, an administrator can also specify the value of product attributes that are used in eligibility determination processing. For more information, see Section 8.3.2, *Configuring Eligibility Determination Processing*. Only published rules will be used when determining eligibility and entitlement within a case.

For detailed information on assigning rules to products, the settings that are used in eligibility determination processing, and publishing rules see the Cúram Integrated Case Management Configuration Guide.

# 8.6 Configuring Services and Referrals

This section provides an overview of the information that is configured for services and referrals in order to create and deliver them at the integrated case level.

Note that the services used by service delivery and referral processing are configured in CPM as service offerings. Service offerings are the templates on which individual services created at the integrated case level are based.

For detailed information on configuring services and referrals, see the Cúram Provider Management Guide.

## 8.6.1 Configuring Services

A number of settings are provided that allow an administrator to configure the information that can be recorded by a case worker when creating and managing services that are based on a particular service offering. Settings are provided that dictate if the service is appropriate for multiple clients and if the automatic creation of multiple service deliveries is allowed. Individual settings also exist on the service offering that dictate if a provider/provider type is mandatory or optional when creating a service. For more information on this setting, see Section 8.6.1.1, *Configuring Service Provider Selection*.

Administrators can also define if a service frequency and client participation

frequency and duration can be specified on service creation. A default service frequency can also be specified and this frequency can be manually overridden by a case worker.

Additionally, a number of settings that dictate the delivery and the financial processing for the service are provided. An administrator can specify if a case worker must record the required number of units of a service that must be delivered to a client. A setting is also provided that dictates if a case worker can specify an authorized rate. If this is set, a case worker can manually override the rate to be paid to a provider for providing the service. Administrators can also define the delivery type for the service which dictates what financial processing will be used.

#### Configuring Service Provider Selection

The following table describes the settings that are available that determine how a service provider is selected at the case level:

Setting	Description
Provider Mandatory	The case worker must select a provider. The case worker is not given the option to select a provider type.
Provider Type Mandatory	The case worker must select a provider type. The caseworker is not given the option to select a provider.
Provider or Provider Type Mandatory	The case worker is given the option to select either a provider or a pro- vider type. Either a provider or a pro- vider type must be selected.
Provider or Provider Type Not Required	The case worker is not presented with the option to select a provider or a provider type.
None Specified	The case worker is given the option to select a provider or a provider type. Provider or provider type are optional.

Table 8.1 Options Available for Configuring Provider and Provider Type Selection

#### Configuring Service Enquiries

The application provides two system administration properties that dictate whether or not email notifications should be sent in respect of service enquiries.

• A system administration property curam.serviceenquiry.SendProviderEmailNotificati

ons allows an administrator to dictate whether or not email notifications are sent to a provider when a service enquiry is created or updated that relates to one of its services.

#### The

curam.serviceenquiry.sendCaseWorkerEmailNotifica tions property allows an administrator to dictate if email notifications are sent to the case worker who owns the service enquiry when the service enquiry is responded to by a provider.

#### 8.6.2 Configuring Referral Services

Referral service configuration includes defining the information that can be specified when creating the referral such as whether the referral service is appropriate for multiple clients

Additionally, the notification templates that are used to produce the referral notifications issued to the client and the provider are configurable. The application also provides a strategy that determines the notification mechanism that is used by the system to send notification letters automatically based on the participants' preferred communication methods and communication exceptions if they exist

#### **Configuring Notification Templates**

The templates on which referral notification letters are based are located in service administration. However, if a template has not been configured, a default pro forma template is provided and used for letter creation. Default templates are located in within system administration. New templates can be defined here and subsequently associated with the service as part of CPM service administration

#### Configuring Follow Up Notifications

The application provides two system administration properties that dictate if and when follow up notifications are sent by the system.

- The property curam.referral.FollowUpApplicableForOrganization allows an administrator to dictate if follow up is required for referrals.
- The property curam.referral.SendEmailNotifications allows an administrator to dictate if email notifications are sent to clients and providers.
- The property curam.referral.EscalationFollowUpWarningDays allows an administrator indicate the number of days after which a supervisor notification should be sent.

#### Configuring a Notification Mechanism

The application provides a strategy that is used by the system to determine the means by which notification letters are issued to providers and clients. The strategy works in conjunction with any communication exceptions and preferred communications recorded for the provider and/or client during registration.

If hard copy and email are communication exceptions, the client and provider cannot be automatically notified. If a provider or client does not have any communication exceptions, the means by which the system sends the notification letter depends on the client and/or provider's preferred communication method.

If email is the preferred communication method and an email address is recorded for the client and provider, the letters are attached to emails and sent. If an email address is not recorded, or hard copy is the preferred communication method, the system sends the letter to a printer for posting. The default printer that is specified for a user in administration is used as the target printer. For information on user administration, see the Cúram Organization Administration Guide. Note that if there are multiple clients on the referral, each client will receive a notification.

If no preferred communication is specified for the provider and client and both address and email details exist, the application provides two system administration properties that allow agencies to specify whether the client and provider letters should be sent either by email or sent to a printer for posting.

- The property curam.referral.DefaultClientCommunicationMethod dictates whether provider letters should be sent by email or sent to a printer for posting.
- The property curam.referral.DefaultProviderCommunicationMetho d dictates whether the client letters should be sent either by email or sent to a printer for posting.

# 8.7 Configuring Case Ownership

The application provides an initial case ownership strategy that can be overridden by the organization as required. An organization can override the default case ownership strategy as required depending upon its requirements to assign case ownership to any user, organization unit, position, or work queue. An administrator can also define a case ownership strategy on a caseby-case basis using workflow.

A Case Ownership Strategy setting is provided that allows an administrator to define a case ownership strategy for cases based on a particular type us-

ing workflow. If an ownership strategy is defined for a case type, this setting defines how the initial case owner for the case should be determined. For example, an administrator may specify that a particular type of integrated case be assigned to a specific work queue

For information on assigning case ownership to users and organization groups, see the Cúram Organization Administration Guide.

For further information on the default case ownership strategy provided by the application and configuring case ownership, see the Cúram Integrated Case Management Configuration Guide.

## 8.8 Configuring Case Milestones

Milestones are used to track the completion of significant events or tasks during the life cycle of a case, for example, to track the submission, approval, and activation of a case. All case milestones are based on an associated milestone configuration which is configured as part of administration. To support the manual and automatic creation of milestones within a case, it is also necessary to set up an association between a milestone configuration and the case type (product delivery, integrated case, or screening type).

Each milestone has an expected start and end date during which the important event is scheduled to start and complete and an actual start and end date. Out-of-the-box, Cúram Integrated Case Management provides functionality to set up milestones. Note, however, some development is required to create milestone deliveries at the case level.

For milestones that are automatically created by the system, the expected start date and expected end date may not be changed unless the milestone is configured to allow the dates to be changed. A milestone may also be configured to require that a milestone waiver request be submitted for approval before the expected start date and expected end date may be changed.

Milestone waiver request approval check settings determine the percentage of submitted waiver requests for a milestone of a particular type that need to be reviewed by a case supervisor. For example, an approval check can be set up on a milestone that requires 60% of all submitted requests to be approved; 40% will not require approval.

Milestone waiver request approval checks can be set up at the organization and user level, for all milestone types or for a milestone of a particular type. User configuration settings taking precedence over organization unit settings. In other words, when a waiver request is submitted for approval by a user, the system first checks the user's milestone waiver request approval check settings, then checks the milestone waiver request approval check settings for the organization unit that the user belongs to. The system may determine at any point in this process that the milestone waiver request requires approval.

For detailed information on case milestone settings, see the Cúram Integrated Case Management Configuration Guide.

# 8.9 Configuring Ongoing Case Management

The application provides a number of settings that allow agencies to configure ongoing case management operations. These settings include setting properties that allow the agency to control the number of entries that appear in the case transaction log case and properties that dictate recent case actions, and case closure.

As part of ongoing case management configuration, the application also provides functionality that allows the agency to configure the case search and case approval. Case search configuration allows the agency to control the type of search that is used when searching for cases at the case level: standard or enhanced.

Case approval can be configured using case approval checks. Administrators can also configure the automatic approval of cases by user role. Approval checks determine the percentage of cases or services that will be reviewed by a case supervisor. For example, a case approval check can be set up such that 50% of cases submitted for approval will require manual approval; the other 50% will be automatically approved. At the organization unit and user levels, it must be specified whether the case approval check should apply to a single product or to all products. Setting it at the service level governs all service delivering that service (across all users).

For more information on configuring ongoing case management, see the Cúram Integrated Case Management Configuration Guide.

# Chapter 9

# **Securing Cases**

#### 9.1 Introduction

Case security allows the agency to secure case information. When a case worker attempts to access case information, case participant information, or perform operations on a case, the system checks the case worker's security access to ensure that he or she has the appropriate user rights. A user's security access is set up as part of security administration. The user's security rights determine what case information can be accessed, viewed, and changed by users. Certain users may not have the same access as other users. For example, case supervisor users may be permitted to approve cases and a trainee case worker user may be restricted from approving cases because of his or her lack of experience. Case security is provided at various levels within the application.

Each level is geared towards securing a particular aspect of a case. The organization can use one or a combination of these levels of security to enable or restrict a user's ability to perform the following activities:

- Perform general case operations
- Accessing fields
- Perform operations on specific programs and features
- Access case and client information based on location
- Access sensitive case information

This chapter provides an overview of each aspect of a case that can be secured using the levels of security available.

# 9.2 Securing General Case Operations

A user's ability to perform general case operations is set up as part of system

administration using function security. General case operations that are accessible from the client include registering a person and creating a case, approving a case, and suspending payments.

Function security governs user access to all functions accessible from the client application. A user's ability to perform case operations is governed by the user's security role, security group, and the security identifiers assigned to the operations during system administration. This data is set up in a hierarchy. The user's security role is the top level of the hierarchy, the user's security group is at the second level, the security identifiers and case operations, which are defined by function identifiers, are at the third and fourth levels respectively.

The hierarchy includes the protected resources that a user can access such as functions, fields, products, assessments, outcome plans, investigations, and appeals.

For more information on securing user access to all server functions, see the Cúram System Configuration Guide.

# 9.3 Protecting Fields from General User Access

Fields can be protected from general user access using field security. Field level security determines whether users can view information in specific fields. For example, the organization may wish to restrict access to a client's salary details from junior case workers. The client's salary details can be protected from general user access.

For more information on protecting fields from general user access, see the Cúram System Configuration Guide.

# 9.4 Securing Case and Client Information Based on Location

Access to case and client information can be secured based on the user's location and the location of the user's position in the organization.

Location based security is used to authorize a user's ability to access case and client information based on his or her location within the organization. A user can access case information only if the user's location is successfully compared to either the location of the case owner if the case owner is a user or the location of the primary client of the case.

For more information on location based security, see the Cúram Location Administration Guide.

# 9.5 Securing Operations for Specific Programs and Features

The application provides a level of security that allows the organization to secure case operations for specific programs. Program security authorizes a user to approve, maintain, create, and/or view program information. The purpose of program security is to provide an extra layer of security for cases that facilitate the authorization and delivery of benefits. For example, an organization may wish to restrict user access to certain types of programs to only a certain set of users in the location that also have the required security identifiers.

Program security is set up on a product by product basis. When a user attempts to perform certain processes associated with a product delivery case, the system determines whether or not the user has the necessary security identifiers for the secured program. This additional layer of security is also available for additional case functionality such as services, assessments, investigations, appeals, payment corrections, and outcome plans.

## 9.6 Securing Access to Sensitive Case Information

Sensitivity security determines user access to sensitive information. This is done by comparing the user's sensitivity level to the element's sensitivity level (sensitivity levels range from 1-5 with 5 being the highest level). Sensitivity ensures that only users who can be trusted with certain information can access that information. For example, certain cases may be high profile. The data on such cases would have a high sensitivity level. Only users with the same high sensitivity level could access the case information.

Case notes are assigned a sensitivity level whereby only users with a sensitivity equal to or higher than the sensitivity level for the case note can view the note details. Additional sensitivity security is available for services and outcome plans: all services, and the actions and attachments within an outcome plan are assigned sensitivity levels. In order for a user to have access to the specific service or action and attachment within an outcome plan, the user must have a sensitivity level equal to or higher than the specific data.

For more information on location based security, see the Cúram Location Administration Guide.

# Chapter 10

# Conclusion

## 10.1 Summary

The following is a summary of the main concepts covered in this guide:

- Cúram Integrated Case Management enables agencies to meet the needs
  of clients and their families in numerous ways until the ultimate goal of
  client self-sufficiency is reached.
- Cúram Integrated Case Management meets the needs of clients in two main ways: through the delivery of benefits using product delivery cases and though the delivery of services and referrals to services.
- The case worker workspace provides tools that help case workers to efficiently balance their work and quickly access important case information
- Ongoing case management tools allow case workers to manage case participants, perform audits on cases, manually change case ownership, and track important issues and legal proceedings regarding case participants.
- Many aspects of ICM are fully configurable. This includes the products, evidence, and rules that determine client eligibility and entitlement as well as services and referrals to services.
- Case security is used to secure access to important case and client information.

### 10.2 Additional Information

Additional information on the topics covered in this guide are covered in several related documents:

Cúram Temporal Evidence Guide

This guide covers the basic concepts of temporal evidence functionality.

#### Cúram Outcome Management Guide

This guide covers the basic concepts of outcome management.

#### Cúram Financials Guide

This guide covers the basic concepts of financial processing.

#### Cúram Provider Management Guide

This guide covers the basic concepts of provider management functionality.

#### Cúram Appeals Guide

This guide covers the basic concepts of appeals functionality.

#### Cúram Calendaring Guide

This guide covers the basic concepts of calendaring functionality.

#### Cúram Deductions Guide

This guide covers the basic concepts of deductions functionality.

#### Cúram Verifications Guide

This guide covers the basic concepts of verifications functionality.

# Cúram Integrated Case Management Configuration Guide

This guide describes the configuration options available within Cúram Integrated Case Management.

#### Working with Cúram Express Rules

This guide provides step by step instructions on how to create Cúram Express Rule rule sets and how to use the Cúram Express Rules Editor to add business and technical logic to a rule set.

#### How to Build a Product

This guide provides sample-based instructions on how to build a product. It starts off with a simple product sample, and builds upon complexity using varying product samples.

#### Cúram Dynamic Evidence Configuration Guide

This guide describes the configuration options available within Cúram Dynamic Evidence.

#### Cúram System Configuration Guide

This guide describes the basic concepts of system configuration.

## **Notices**

This information was developed for products and services offered in the U.S.A. IBM may not offer the products, services, or features discussed in this document in other countries. Consult your local IBM representative for information on the products and services currently available in your area. Any reference to an IBM product, program, or service is not intended to state or imply that only that IBM product, program, or service may be used. Any functionally equivalent product, program, or service that does not infringe any IBM intellectual property right may be used instead. However, it is the user's responsibility to evaluate and verify the operation of any non-IBM product, program, or service. IBM may have patents or pending patent applications covering subject matter described in this document. The furnishing of this document does not grant you any license to these patents. You can send license inquiries, in writing, to:

IBM Director of Licensing

**IBM Corporation** 

North Castle Drive

Armonk, NY 10504-1785

U.S.A.

For license inquiries regarding double-byte (DBCS) information, contact the IBM Intellectual Property Department in your country or send inquiries, in writing, to:

**Intellectual Property Licensing** 

Legal and Intellectual Property Law.

IBM Japan Ltd.

1623-14, Shimotsuruma, Yamato-shi

Kanagawa 242-8502 Japan

The following paragraph does not apply to the United Kingdom or any other country where such provisions are inconsistent with local law: INTERNATIONAL BUSINESS MACHINES CORPORATION PROVIDES THIS PUBLICATION "AS IS" WITHOUT WARRANTY OF ANY KIND, EITHER EXPRESS OR IMPLIED, INCLUDING, BUT NOT LIMITED TO, THE IMPLIED WARRANTIES OF NON-INFRINGEMENT, MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE. Some states do not allow disclaimer of express or implied warranties in certain transactions, therefore, this statement may not apply to you.

This information could include technical inaccuracies or typograph-

ical errors. Changes are periodically made to the information herein; these changes will be incorporated in new editions of the publication. IBM may make improvements and/or changes in the product(s) and/or the program(s) described in this publication at any time without notice.

Any references in this information to non-IBM Web sites are provided for convenience only and do not in any manner serve as an endorsement of those Web sites. The materials at those Web sites are not part of the materials for this IBM product and use of those Web sites is at your own risk.

IBM may use or distribute any of the information you supply in any way it believes appropriate without incurring any obligation to you. Licensees of this program who wish to have information about it for the purpose of enabling: (i) the exchange of information between independently created programs and other programs (including this one) and (ii) the mutual use of the information which has been exchanged, should contact:

**IBM Corporation** 

Dept F6, Bldg 1

294 Route 100

Somers NY 10589-3216

U.S.A.

Such information may be available, subject to appropriate terms and conditions, including in some cases, payment of a fee.

The licensed program described in this document and all licensed material available for it are provided by IBM under terms of the IBM Customer Agreement, IBM International Program License Agreement or any equivalent agreement between us.

Any performance data contained herein was determined in a controlled environment. Therefore, the results obtained in other operating environments may vary significantly. Some measurements may have been made on development-level systems and there is no guarantee that these measurements will be the same on generally available systems. Furthermore, some measurements may have been estimated through extrapolation. Actual results may vary. Users of this document should verify the applicable data for their specific environment.

Information concerning non-IBM products was obtained from the suppliers of those products, their published announcements or other publicly available sources.

IBM has not tested those products and cannot confirm the accuracy of performance, compatibility or any other claims related to non-IBM products. Questions on the capabilities of non-IBM products should be addressed to the suppliers of those products.

All statements regarding IBM's future direction or intent are subject to change or withdrawal without notice, and represent goals and objectives only

All IBM prices shown are IBM's suggested retail prices, are current and are subject to change without notice. Dealer prices may vary.

This information is for planning purposes only. The information herein is subject to change before the products described become available.

This information contains examples of data and reports used in daily business operations. To illustrate them as completely as possible, the examples include the names of individuals, companies, brands, and products. All of these names are fictitious and any similarity to the names and addresses used by an actual business enterprise is entirely coincidental.

#### **COPYRIGHT LICENSE:**

This information contains sample application programs in source language, which illustrate programming techniques on various operating platforms. You may copy, modify, and distribute these sample programs in any form without payment to IBM, for the purposes of developing, using, marketing or distributing application programs conforming to the application programming interface for the operating platform for which the sample programs are written. These examples have not been thoroughly tested under all conditions. IBM, therefore, cannot guarantee or imply reliability, serviceability, or function of these programs. The sample programs are provided "AS IS", without warranty of any kind. IBM shall not be liable for any damages arising out of your use of the sample programs.

Each copy or any portion of these sample programs or any derivative work, must include a copyright notice as follows:

- © your company name) (year). Portions of this code are derived from IBM Corp. Sample Programs.
- © Copyright IBM Corp. \_enter the year or years\_. All rights reserved.

If you are viewing this information softcopy, the photographs and color illustrations may not appear.

# **Trademarks**

IBM, the IBM logo, and ibm.com are trademarks or registered trade-

marks of International Business Machines Corp., registered in many jurisdictions worldwide. Other product and service names might be trademarks of IBM or other companies. A current list of IBM trademarks is available on the Web at "Copyright and trademark information" at http://www.ibm.com/legal/us/en/copytrade.shtml.

Adobe, the Adobe logo, and Portable Document Format (PDF), are either registered trademarks or trademarks of Adobe Systems Incorporated in the United States, other countries, or both.

Microsoft, Word and Excel are trademarks of Microsoft Corporation in the United States, other countries, or both.

Other names may be trademarks of their respective owners. Other company, product, and service names may be trademarks or service marks of others.